MINISTRY FOR PRIMARY INDUSTRIES

IMPORTING COUNTRIES PHYTOSANITARY REQUIREMENTS

GREAT BRITAIN

Status: Approved

Date: 23 December 2020

EXPORTERS ARE ADVISED TO CONFIRM THE PHYTOSANITARY IMPORT REQUIREMENTS PRIOR TO EXPORT FROM NEW ZEALAND

Amendment No.	Date:	Nature of amendment:	Approved by:
27		Updated note for <i>Humulus lupulus</i> (hop cones) under section 4.2 Fresh Cut Flowers and Foliage.	MLM
	19 September 2023	Updated note for <i>Humulus</i> <i>lupulus</i> (pelleted hops and hop powder) under section 4.6 Miscellaneous.	
		Updated section 2.4 Phytosanitary Certificates to include the following:	
		Phytosanitary certificates must be:	
26 30 Jun	30 June 2023	 Issued within 13 days prior to export or issued within 13 days after export 	GF
		AND	
		 Issued before the consignment arrives in Great Britain. 	

Amendment No.	Date:	Nature of amendment:	Approved by:
		Note: MPI recommends certificates are issued prior to export if means of conveyance is via air.	
		Updated the compliance programme country for <i>Cyathea</i> spp. and <i>Dicksonia</i> spp. under section '4.3.2 Cuttings'	
25	27 June 2023	Added <i>Asplenium</i> spp. under section '4.3.6 Whole Plants'	HVC
		Updated the requirements for <i>Cyathea</i> spp. and <i>Dicksonia</i> spp. under section '4.3.6 Whole Plants'	
24	24 March 2023	Removed the following statement under section 2.4 Phytosanitary Certificates, "Note: Until enforcement from 4 January 2023, if phytosanitary certificates are issued after export, the inspection date must be listed on the phytosanitary certificate." as 4 January 2023 has passed. Added the following statement under section 2.4 Phytosanitary Certificates, "Note: Replacement phytosanitary certificates may be issued after export, provided the following additional declaration is included on the replacement phytosanitary certificate: [Insert original phytosanitary certificate number] WAS ISSUED ON [insert date].	GF
23	28 February 2023	Updated section 2.4 Phytosanitary Certificates, section 3.4 Fresh Cut Flowers and Foliage, and section 4.2 Fresh Cut Flowers and Foliage.	GF
		Updated note for Capsicum spp.	

Amendment No.	Date:	Nature of amendment:	Approved by:
		and <i>Solanum lycopersicum</i> under section 4.4 Seed, Grain and Nuts for Sowing - temporary alternative additional declaration is able to be used until 31 December 2023.	
22	23 February 2023	Removed the following sentence under section 2.3.1 Exemption for prohibited items (Scientific license/Letter of Authority) "To arrange inspection and endorsement of a scientific licenses (Letter of Authority), or for further information, please contact an Independent Verification Agency (IVA)", as a letter of authority is no longer required to be endorsed by MPI for Great Britain.	GF
21	26 January 2023	Added treatment options for <i>Cyathea</i> spp. and <i>Dicksonia squarrosa</i> under section 4.3.2 'Cuttings'	НМ
20	6 December 2022	Removed under section 2.4 'Phytosanitary Certificates', the requirement for phytosanitary inspections to be conducted with 14 days. Updated under section 2.4 'Phytosanitary Certificates' that Phytosanitary certificates are not to be issued after export (enforced from 4 January 2023).	HM
19	4 August 2022	RemovedRNQPMeloidogynearenariaandMeloidogynejavanicafromOleaeuropaeaunder section 4.3.6Whole Plants.Updated additional declaration 14.'Plants of herbaceous species forplanting, other than bulbs, corms,plants of the family Gramineae,rhizomes, seeds, tubers' undersection 4.3.7Additionaldeclarations for nursery stock (i.e.Plants for planting)'.AddedEotetranychussexmaculatusandapicalis,toAppendix2:	НМ

Amendment No.	Date:	Nature of amendment:	Approved by:
		Quarantine Pest List.	
		Added end date for the quarantine pest <i>Thekopsora minima</i> in Appendix 2: Quarantine Pest List as moved to list of GB regulated non-quarantine pests.	
		Updated note for <i>Humulus lupulus</i> (Hop cones) under section 4.2 Fresh Cut Flowers and Foliage.	
18	30 June 2022	Updated requirements for Pelleted hops (Humulus lupulus) and Hop powder (Humulus lupulus) under section 4.6 Miscellaneous.	JR
17	20 June 2022	Added requirements for <i>Gentiana</i> spp. & <i>Gentianella</i> spp. under section 4.3.3 'Tissue Culture'	НМ
16	31 May 2022	Added requirements for <i>Agathis</i> spp. and <i>Cordyline</i> spp. to section 4.3.6 Whole Plants.	НМ
15	06 May 2022	Added additional declaration 6 under <i>Phormium spp.,</i> <i>Pittosporum spp., Pseudopanax</i> <i>spp. Pseudowintera spp.</i> in section 4.3.6 'Whole Plants'. Removed additional declaration 6 under <i>Hibiscus syriacus</i> and <i>Wisteria</i> spp. in section 4.3.6 'Whole Plants'.	НМ
14	1 April 2022	Updated spelling of 'Acacia baileyana purpera' to 'Acacia baileyana purpurea' under section 4.4 Seed, Grain and Nuts for sowing. Added commodity requirements, including regulated non- quarantine pests (RNQPs) thresholds and requirements, for <i>Capsicum</i> spp. and <i>Solanum</i> <i>lycopersicum</i> to section 4.4 Seed, Grain and Nuts for sowing. Updated the conditions reference from 3.4 to 3.7 under section 4.4 Seed, Grain and Nuts for sowing. Added Appendix 3 outlining 'Sampling schemes of seeds,	HM

Amendment No.	Date:	Nature of amendment:	Approved by:
		other than seeds of <i>Capsicum</i> spp. varieties which are known to be resistant to the specified pest'	
13	25 November 2021	Updated requirements for <i>Humulus lupulus</i> (Hop cones) under section 4.2 Fresh Cut Flowers and Foliage. Updated requirements for Pelleted hops and Hop powder (<i>Humulus lupulus</i>) under section 4.6 Miscellaneous.	MLM
12	12 October 2021	Added <i>Polymnia sonchifolia</i> to section 2.2 Prohibitions. Removed phytosanitary certificate requirement for <i>Humulus lupulus</i> (Hop cones) under section 4.2 Fresh Cut Flowers and Foliage. Updated conditions for <i>Olea</i> <i>europaea</i> under section 4.3.6 Whole Plants. Added conditions for <i>Vaccinium</i> spp. under section 4.3.6 Whole Plants. Removed phytosanitary certificate requirement for Pelleted hops (<i>Humulus lupulus</i>) under section 4.6 Miscellaneous. Added requirements for Hop powder (<i>Humulus lupulus</i>) under section 4.6 Miscellaneous.	MLM/GF
11	5 August 2021	Updated additional declarations for <i>Brachyscome</i> spp., <i>Fuchsia</i> spp. and <i>Hibiscus syriacus</i> under Section 4.3.6 Whole Plants. Updated Appendix 1: List of Xylella host plants.	GF
10	2 July 2021	Updated requirements for <i>Humulus lupulus</i> (Hop cones) under section 4.2 Fresh Cut Flowers and Foliage. Updated requirements for Pelleted hops (<i>Humulus lupulus</i>) under section 4.6 Miscellaneous. Amended additional declaration	MLM
	Status: APPROVE	requirement for Sophora spp.	Page 5 of 87

Amendment No.	Date:	Nature of amendment:	Approved by:
		under Section 4.3.6 Whole Plants.	
9	25 June 2021	Updated additional declaration number 11 in Section 4.3.7 Additional declarations for Nursery Stock	GF
		Updated Table 2 heading under Section 2.2 Prohibitions.	
		Added note under Section 2.4 Phytosanitary Certificates, Number of items in consignment, to clarify quantity specification.	
		Added note under Section 3.6 Nursery Stock (i.e. plants for planting).	
		Updated additional declarations under Section 4.3.2 Cuttings.	
8	22 June 2021	Added commodity requirements for <i>Hoya</i> spp. under Sections 4.3.2 Cuttings and 4.3.6 Whole Plants.	MLM
		Updated additional declarations under Section 4.3.3 Tissue Culture, to remove additional declaration number 4, and to specify additional declaration that may be conditional.	
		Updated note statements under Section 4.3.1 Budwood for <i>Malus</i> spp.	
		Amended additional declaration number 11 in Section 4.3.7 Additional declarations for Nursery Stock.	
7	9. kmc 2021	Updated Appendix 2: Quarantine Pest List.	
7	8 June 2021	Updated Additional Declaration 4 under Section 4.3 Nursery Stock.	FA/MLM
6	25 May 2021	Updated section 2.4 Phytosanitary Certificates to include quantity specifications.	GF
	20 May 2021	Updated section 3.2 Dried Fruit and Vegetables.	
		Updated section 3.3 Frozen Fruit	

Amendment No.	Date:	Nature of amendment:	Approved by:
		and Vegetables.	
		Updated section 3.7 Seed, Grain and Nus for Sowing.	
5	13 May 2021	Updated nursery stock additional declarations under Section 4.3.4 Tubers and corms, Section 4.3.5 Rhizomes, Section 4.3.6 Whole Plants to specify requirements which may be conditional.	FA
4	6 May 2021	Updated import requirements for Acer japonicum, Acer palmatum, Acer shirasawanum under Section 4.3.6 Whole Plants. Updated Additional Declaration 17	FA
		and 20.	
		Updated the following additional declarations for <i>Olea europaea</i> and Additional Declaration 19 under Section 4.3 Nursery Stock. Added new section 2.3.1	
3	7 April 2021	Exemption for prohibited items (Scientific license/Letter of Authority).	FA
		Updated Section 2.4 Phytosanitary Certificates to include information regarding phytosanitary certificate requests in ePhyto.	
2	5 February 2021	Updated requirements for <i>Humulus lupulus</i> (Hop cones) under section 4.2 Fresh Cut Flowers and Foliage.	MLM
2		Created a new section 4.6 Miscellaneous and added requirements for pelleted hops (<i>Humulus lupulus</i>).	
1	23 December 2020	New ICPR	GF/FA/MLM

DISCLAIMER

The phytosanitary requirements in this document may be used as the basis for export certification. However, exporters should be aware that importing countries may change their requirements at any time; at short notice or without giving notice to New Zealand.

This information is provided strictly on the basis that the Crown, the Ministry for Primary Industries, its statutory officers, employees, agents and all other persons responsible for or associated with the compilation, writing, editing, approval or publication of the information:

1. disclaim any and all responsibility for any inaccuracy, error, omission, lateness, or any other kind of inadequacy, deficiency or flaw in, or in relation to, the information; and

2. without limiting (1) above, fully exclude any and all liability of any kind on the part of all of them, to any person or entity that chooses to rely on this information.

Compliance with this document is not to be taken as a guarantee that any particular goods will be granted access to any overseas market. We recommend that exporters work with their importers to obtain the most up-to-date information.

Table of Contents

1	Ge 1.1	neral Information For enquiries email the Plant Exports Team:	
	1.2	Scope	11
	1.3	Member Countries of Great Britain	12
	1.4	Phytosanitary Legislation	12
	1.5	Fees and Charges	12
2	Ge	neral Requirements	12
	2.1	Definitions	12
	2.2	Prohibitions	13
	2.3	Import Permits	16
	2.3.1	Exemption for prohibited items (Scientific license/Letter of Authority)	16
	2.4	Phytosanitary Certificates	16
	2.5	Quarantine Pests	18
	2.6	Inspection on Arrival	18
	2.7	Ports of Entry	18
	2.8	Transit Requirements	18
	2.9	Wood Packaging	18
3	Coi 3.1	mmodity Class Requirements Fresh Fruit and Vegetables	
	3.2	Dried Fruit and Vegetables	18
	3.3	Frozen Fruit and Vegetables	18
	3.4	Fresh cut flowers and foliage	19
	3.5	Dried Cut Flowers and Foliage	19
	3.6	Nursery Stock (i.e. plants for planting)	19
	3.7	Seed, Grain and Nuts for human consumption	19
	3.8	Seed, Grain and Nuts for Sowing	19
	3.9	Growing Media and Packing Material	19
4	Coi 4.1	mmodity Specific Requirements Fresh fruit and Vegetables	
	4.2	Fresh Cut Flowers and Foliage	
	4.3	Nursery Stock (i.e. plants for planting)	
		Budwood	
		Cuttings	
		Tissue Culture	
		Tubers and corms	
		Rhizomes	
		Whole Plants	
I	CPR GR	REAT BRITAIN Status: APPROVED <u>19 September 2023</u> Page 9 G	of 87

4.3.7	Additic	nal declarations for nursery stock (i.e. Plants for planting)		
4.4	Seed,	Grain and Nuts for sowing		
4.4.2 \$	4.4.2 Seeds, Grains and Nuts for Processing68			
4.5	Growir	ng media	68	
4.6	Miscel	laneous	68	
Appen	ndix 1:	List of Xylella host plants	70	
Appen	ndix 2:	Quarantine Pest List	72	
Appendix 3: Sampling schemes of seeds for Capsicum spp. and Solanum lycopersicum,				

other than seeds of *Capsicum* spp. varieties which are known to be resistant to the specified pest: 86

1 General Information

As of 1 January 2021, this Plants ICPR comes into force.

Note: This ICPR applies to England, Scotland and Wales only. For Northern Ireland requirements, please refer to the European Union ICPR.

Users of this document are strongly advised to read all sections to understand the phytosanitary requirements for a commodity.

1.1 For enquiries email the Plant Exports Team: plantexports@mpi.govt.nz

To help the Plant Exports Team process your email please record in the subject line of your email descriptive keywords which are relevant to your enquiry i.e. ICPR enquiry-Country-specific details.

For urgent enquiries phone the Plant Exports Team. Telephone: +64 4 894 5693

1.2 Scope

The requirements listed in this ICPR apply to product of New Zealand only, unless specifically stated.

As there are no New Zealand legislative requirements for certification of exported plant products, the technical phytosanitary requirements are determined by the importing country for plant produce being imported into their country. In this respect, ICPRs are MPI's summary of the importing country's legal requirements and thus forms a basis upon which export phytosanitary certification is provided.

Where an importing country operates on the basis of issuing import permits to their importers, the import permit conditions take precedence over any technical conditions contained in the MPI ICPR for that country.

This ICPR specifies the Great Britain's phytosanitary requirements for the stated commodities and commodity classes only. If a commodity or commodity grouping is not identified within Section 3 or Section 4 exporters should contact:

• The Great Britain directly to ascertain requirements

or

Ministry for Primary Industries (MPI) – Plant Exports

Users of this document are strongly advised to review all sections of the ICPR for the determination of a commodity's phytosanitary requirements.

1.3 Member Countries of Great Britain

Note: Phytosanitary requirements for Northern Ireland are listed in the European Union ICPR here: <u>https://www.mpi.govt.nz/legal/compliance-requirements/icpr-importing-countries-phytosanitary-requirements/icprs-by-country/</u>

1.4 Phytosanitary Legislation

The following legislation controls the importation of plants and plant materials into the Great Britain:

 The Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020.

1.5 Fees and Charges

Please note that the determination and provision of phytosanitary requirements for a commodity not listed within the ICPR may be undertaken on a cost recovered basis. A link to the list of Plant Exports Fees and Charges is available on http://mpi.govt.nz/exporting/food/fruit-and-vegetables/fees-and-charges/

2 General Requirements

2.1 Definitions

Table 1. Definition of important glossaries.

Plants	Living plants and specified living parts thereof, including seeds. Living parts of plants include: - fruit, in the botanical sense, other than that preserved by deep freezing - vegetables, other than those preserved by deep freezing - tubers, corms, bulbs, rhizomes - cut flowers - cut flowers - branches with or without foliage - cut trees retaining foliage - leaves and foliage - plant tissue cultures - live pollen - budwood, cuttings, scions
Seeds	Seeds in the botanical sense, other than those
ICPR GREAT BRITAIN	Status: APPROVED 19 September 2023 Page 12 of 87

	not intended for planting.
Plant products	Products of plant origin, unprocessed or having undergone simple preparation, in so far as these are not plants.
Planting	Any operation for the placing of plants to ensure their subsequent growth, reproduction or propagation.
Plants intended for planting	Plants which are already planted and are intended to remain planted or to be replanted after their introduction or plants which are not planted at the time of introduction, but are intended to be planted thereafter.
Harmful organisms	Pests of plants or of plant products, which belong to the animal or plant kingdoms, or which are viruses, mycoplasmas or other pathogens.

2.2 **Prohibitions**

Table 2. List of commodities prohibited for entry into	Great Britain.
--	----------------

Scientific Name	Common Name	Prohibited Plants or
		plant parts
Abies spp.	Fir	Plants, other than fruits or
		seeds.
Castanea spp.	Chestnut	Plants with leaves, other
		than fruits or seeds.
<i>Cedrus</i> spp.	Cedar	Plants, other than fruit or
		seeds.
Chaenomeles spp.		Plants for planting, other
		than dormant plants free
		from leaves, flowers and
		fruits
Chamaecyparis spp.	Cypress	Plants, other than fruit or
		seeds.
<i>Citrus</i> spp.		Plants for planting, other
(and their hybrids)		than seeds
<i>Crataegus</i> spp.		Plants for planting, other
		than dormant plants free
		from leaves, flowers and
		fruits
<i>Cydonia</i> spp.		Plants for planting, other
		than dormant plants free
		from leaves, flowers and
		fruits
Fortunella spp.	Kumquat	Plants for planting, other
(and their hybrids)		than seeds
<i>Juniperus</i> spp.	Juniper	Plants, other than fruit or
		seeds.
<i>Larix</i> spp.	Larch	Plants, other than fruit or
		seeds.
ICPR GREAT BRITAIN	Status: APPROVED	19 September 2023 Page 13 of 87

Malus spp.		Plants for planting, other
		than dormant plants free
		from leaves, flowers and
		fruits
Picea spp.	Spruce	Plants, other than fruit or
7 1000 3pp.	Oprace	seeds.
Pinus spp.	Pine	Plants, other than fruit or
		seeds.
Poncirus spp.	Ornamental citrus	Plants for planting, other
(and their hybrids)		than seeds
Poaceae		Plants for planting, other
		than seeds and other than
		plants of ornamental
		perennial grasses of the
		subfamilies
		Bambusoideae and
		Panicoideae and of the
		genera <i>Buchloe,</i>
		Bouteloua,
		Calamagrostis,
		Cortaderia, Glyceria,
		Hakonechloa, Hystrix,
		Molinia, Phalaris,
		Shibataea, Spartina, Stipa
		and Uniola
Prunus spp.		Plants for planting, other
		than dormant plants free
		from leaves, flowers and
		fruits
<i>Pseudotsuga</i> spp.	Douglas fir	Plants, other than fruit or
_		seeds.
<i>Pyrus</i> spp.		Plants for planting, other
		than dormant plants free
		from leaves, flowers and
		fruits
Quercus spp.		Plants with leaves, other
Descent		than fruit or seeds.
Rosa spp.		Plants for planting, other
		than dormant plants free
		from leaves, flowers and
Soil		fruits
Soil		Consisting in part of solid
Growing modium	organic substances	
Growing medium		Consisting in whole or in part of solid organic
		substances, other than
		any composed entirely of
		peat or fibre of Cocos
		nucifera, previously not
		used for growing of plants
		ased for growing or plants

		or for any agricultural purposes
Solanaceae	Various	Tubers of Solanum tuberosum, seed potatoes.
		Plants for planting of stolon or tuber forming species of <i>Solanum</i> spp. and their hybrids other than those tubers of <i>Solanum tuberosum</i>
		Plants of <i>Solanaceae</i> intended for planting, other than seeds
<i>Tsuga</i> spp.	Hemlock	Plants, other than fruit or seeds.
Vitis spp.	Grape	Plants, other than fruit.

List of high-risk plants, plant products and other objects from third countries which may not be introduced into Great Britain pending a risk assessment

1. Plants for planting, other than seeds, *in vitro* material and naturally or artificially dwarfed woody plants for planting, and belonging to the following genera or species:

<i>Acacia</i> spp.	<i>Albizia</i> spp.	Alnus spp.
<i>Annona</i> spp.	<i>Bauhinia</i> spp.	Berberis spp.
<i>Betula</i> spp.	<i>Caesalpinia</i> spp.	<i>Cassia</i> spp.
Castanea spp.	Cornus spp.	Corylus spp.
<i>Crataegus</i> spp.	<i>Diospyros</i> spp.	<i>Fagu</i> s spp.
Ficus carica	<i>Fraxinus</i> spp.	Hamamelis spp.
<i>Jasminum</i> spp.	<i>Juglans</i> spp.	<i>Ligustrum</i> spp.
<i>Lonicera</i> spp.	<i>Malus</i> spp.	<i>Nerium</i> spp.
Persea spp.	Polymnia sonchifolia	Populus spp.
	(applies as of 2	
	December 2021)	
<i>Prunus</i> spp.	Quercus spp.	<i>Robinia</i> spp.
<i>Salix</i> spp.	Sorbus spp.	<i>Taxus</i> spp.
<i>Tilia</i> spp.	<i>Ulmu</i> s spp.	

- 2. Plants of *Ullucus tuberosus*
 - 3. Plants, other than fruit or seeds of *Coffea* spp. or *Polygala myrtifolia* which originate in any third country.

2.3 Import Permits

Import permits state the phytosanitary requirements for importation.

Import permits are required for PROHIBITED commodities for scientific research purposes or an exemption ('derogation') to the rules.

Permits may be requested from: <u>https://www.gov.uk/guidance/moving-prohibited-plants-plant-pests-pathogens-and-soil</u>

2.3.1 Exemption for prohibited items (Scientific license/Letter of Authority)

You must get a licence from the Animal and Plant Health Agency (APHA) to import, move or keep prohibited:

- plants, including parts of plants and seeds
- invertebrate plant pests (arthropods, molluscs and nematodes)
- plant pathogens (fungi, bacteria, viruses, virus-like agents and phytoplasmas)
- soil and other organic material
- potatoes

Your licence will include the specific conditions under which you must import, move or keep the material.

You can only get a licence if you're using the prohibited items for:

- official testing
- scientific research
- educational purposes
- trials
- testing new varieties of plants (varietal selection)
- breeding

For information on obtaining scientific licenses (Letter of Authority): <u>https://www.gov.uk/guidance/moving-prohibited-plants-plant-pests-pathogens-and-soil</u>

2.4 Phytosanitary Certificates

Phytosanitary certificates are required to accompany 'controlled' consignments of:

- Fresh cut flowers and foliage (except *Humulus lupulus* (hop cones))
- Fresh fruit and vegetables
- Nursery stock (i.e. plants for planting)
- Seed for sowing

'Controlled' material is outlined in section 4 Commodity Specific Requirements.

For all phytosanitary certificate requests in ePhyto for plants and plant product exports to Great Britain (England, Scotland and Wales), please select country of destination as "United Kingdom".

Quantities of plant and plant products listed on the phytosanitary certificate must be recorded using the following specified units:

Kilograms (kg):

 Fruit, vegetables, soil and branches with foliage, potatoes, grain, bulbs, corms and rhizomes, plants in tissue culture and seeds

Number of stems:

- Cut flowers

Number of items in consignment:

- Any other plants for planting **Note:** Items can be number of plants

The requirements for phytosanitary certificate issuance are as follows:

Phytosanitary certificates must be:

1. Issued within 13 days prior to export or issued within 13 days after export

AND

2. Issued before the consignment arrives in Great Britain.

Note: MPI recommends certificates are issued prior to export if means of conveyance is via air.

All postal consignments must have the original phytosanitary certificate attached to the outside of the package on entry into Great Britain.

A plant passport may be required for material moving within the Great Britain. Please note that this is not a phytosanitary certificate.

Information on plant passports: <u>https://www.gov.uk/guidance/issuing-plant-passports-to-trade-plants-in-the-eu</u>

The following is a list of plants and plant products which do not require a phytosanitary certificate:

- Fruits of Ananas comosus
- Fruits of Actinidia spp.
- Fruits of Cocos nucifera
- Fruit and leaves of *Citrus* spp.
- Fruit of *Fortunella* spp.
- Fruit of *Poncirus* spp.
- Fruit of *Diospyros* spp.
- Fruits of Durio zibethinus
- Fruits (bolls) of Gossypium spp.
- Leaves of Murraya spp.
- Fruits of *Musa* spp.
- Fruits of *Mangifera* spp.

- Fruits of Phoenix dactylifera
- Fruits of *Passiflora* spp.
- Fruits of *Psidium* spp.

2.5 Quarantine Pests

Refer to Appendix 2.

2.6 Inspection on Arrival

All consignments are subject to inspection on arrival into the Great Britain.

2.7 Ports of Entry

Not specified.

2.8 Transit Requirements

Requirements not specified.

2.9 Wood Packaging

Refer to forestry ICPR for Great Britain, link below:

Forestry Importing Countries Phytosanitary Requirements page: <u>https://www.mpi.govt.nz/law-and-policy/requirements/icpr-importing-countries-phytosanitary-requirements/forestry-icprs/</u>.

3 Commodity Class Requirements

3.1 Fresh Fruit and Vegetables

Conditions:

Import permit not required. Phytosanitary certificate may be required. Refer to section 2.4 phytosanitary certificates and section 4.1 commodity specific requirements.

Fruit must be free from leaves and peduncles.

3.2 Dried Fruit and Vegetables

Import permit not required. Phytosanitary certificate not required.

3.3 Frozen Fruit and Vegetables

Import permit not required. Phytosanitary certificate not required.

3.4 Fresh cut flowers and foliage

Conditions:

Import permit not required. Phytosanitary certificate required (except *Humulus lupulus* (hop cones)). Refer to section 2.4 phytosanitary certificates and section 4.2 commodity specific requirements.

3.5 Dried Cut Flowers and Foliage

Requirements not specified.

3.6 Nursery Stock (i.e. plants for planting)

Conditions:

Import permit not required. Phytosanitary certificate required.

Refer to section 4.3 commodity specific requirements.

Note: Refer to section 4.3.7 Additional declarations for Nursery Stock, for additional declarations which may apply.

3.7 Seed, Grain and Nuts for human consumption

Requirements not specified.

3.8 Seed, Grain and Nuts for Sowing

Conditions:

Import permit not required. Phytosanitary certificate required. Refer to section 2.4 phytosanitary certificates and section 4.4 for commodity specific requirements.

3.9 Growing Media and Packing Material

Conditions:

PROHIBITED, other than composed entirely of peat (whole peat).

Whole peat

Conditions:

Import permit not required. Phytosanitary certificate required.

4 Commodity Specific Requirements

4.1 Fresh fruit and Vegetables

Conditions: Refer to section 3.1.

<u>Actinidia deliciosa</u>

Conditions:

<u>Kiwifruit</u>

Import permit not required. Phytosanitary certificate not required.

<u>Allium cepa</u>

Conditions:

Import permit not required. Phytosanitary certificate required.

<u>Citrus spp.</u> Conditions: Lemon, lime, orange etc.

Squash, pumpkin, ahourd

Import permit not required. Phytosanitary certificate not required.

Import permit not required. Phytosanitary certificate required.

<u>Cucurbita spp.</u> Conditions:

Malus spp.

<u>Apple</u>

Onion

Conditions:

Import permit not required. Phytosanitary certificate and Additional declaration required.

Additional declaration:

"Consignment complies with Annex 7, Part A, point 97, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the fruits originate in a country which, in accordance with the measures specified in ISPM 4, is known to be free from *Botryosphaeria kuwatsukai* (Hara) G.Y. Sun and E. Tanaka"

ÀND

"Consignment complies with Annex 7, Part A, point 98, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the fruits originate in a country which, in accordance with the measures specified in ISPM 4, is known to be free from *Anthonomus quadrigibbus* Say"

AND

"Consignment complies with Annex 7, Part A, point 99, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the fruits originate in a country which, in accordance with the measures specified in ISPM 4, is known to be free from *Grapholita prunivora* (Walsh), *Grapholita inopinata* (Heinrich) and *Rhagoletis pomonella* (Walsh)"

Note: New Zealand is known to be free from *Botryosphaeria kuwatsukai* (Hara) G.Y Sun and E. Tanaka, *Anthonomus quadrigibbus* (Say), *Grapholita prunivora* (Walsh), *Grapholita inopinata* (Heinrich), and *Rhagoletis pomonella* (Walsh).

<u>Prunus spp.</u>

Cherry, Apricot, Peach, Plum

Conditions:

Import permit not required. Phytosanitary certificate required.

Pyrus spp.

<u>Pear</u>

Conditions:

Import permit not required. Phytosanitary certificate and Additional declaration required.

Additional declaration:

"Consignment complies with Annex 7, Part A, point 97, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the fruits originate in a country which, in accordance with the measures specified in ISPM 4, is known to be free from *Botryosphaeria kuwatsukai* (Hara) G.Y. Sun and E. Tanaka"

AND

"Consignment complies with Annex 7, Part A, point 98, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the fruits originate in a country which, in accordance with the measures specified in ISPM 4, is known to be free from *Anthonomus quadrigibbus* Say"

Note: New Zealand is free from *Botryosphaeria kuwatsukai* (Hara) G.Y Sun and E. Tanaka, *Anthonomus quadrigibbus* (Say), *Grapholita prunivora* (Walsh), *Grapholita inopinata* (Heinrich), and *Rhagoletis pomonella* (Walsh).

All other fruits and vegetables

Conditions:

Requirements not specified. Contact PlantExports@mpi.govt.nz.

4.2 Fresh Cut Flowers and Foliage

Conditions:

Refer Section 3.2.1.

Import permit not required. Phytosanitary certificate required. Additional declaration may be required. Contact <u>PlantExports@mpi.govt.nz</u>.

<u>Humulus lupulus</u>

Hop cones

Conditions:

Import permit not required. Phytosanitary certificate not required. Attestation certificate of equivalence required.

Note: Hops imported into GB until 1 February 2025, are required to be accompanied by an EU Attestation of Equivalence.

For further information please see:

Commission Regulation (EC) No 1295/2008 of 18 December 2008 on the importation of hops from third countries (Codified version)

Note: Annex II of this regulation is a template of an EU Attestation certificate of equivalence.

All other cut flowers and foliage

Conditions:

Requirements not specified. Contact <u>PlantExports@mpi.govt.nz</u>.

4.3 Nursery Stock (i.e. plants for planting)

4.3.1 Budwood

<u>Malus spp.</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 10, 11

AND

Note: The additional declarations below apply to any third country where Cherry rasp leaf virus is known to occur.

"Consignment complies with Annex 7, Part A, point 73, option (a), item (i) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants have been officially certified under a certification scheme requiring them to be derived in direct line from material which has been maintained under appropriate conditions and has been subjected to official testing for at least Cherry rasp leaf virus using appropriate indicators or equivalent methods and has been found free from the pests tested"

OR

"Consignment complies with Annex 7, Part A, point 73, option (a), item (ii) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants have been derived in direct line from material which has been maintained under appropriate conditions and has been subjected, at least once within the last three complete cycles of vegetation, to official testing for at least Cherry rasp leaf virus using appropriate indicators or equivalent methods and has been found free from the pests tested"

AND

"Consignment complies with Annex 7, Part A, point 73, option (b) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that no symptoms of diseases caused by Cherry rasp leaf virus have been observed on plants at the place of production, or on susceptible plants in its immediate vicinity, since the beginning of the last complete cycle of vegetation" AND

Note: The additional declarations below apply to any third country where *Candidatus Phytoplasma* 'mali' Seemüller & Schneider is known to occur

Note: Where * is indicated below, the name of the area(s) must be included in the phytosanitary certificate under the heading "Additional declaration"

"Consignment complies with Annex 7, Part A, point 74, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants originate in an area* which, in accordance with the measures specified in ISPM4, is known to be free from Candidatus Phytoplasma 'mali' Seemüller & Schneider"

OR

Note: Applies to plants, other than plants raised from seeds

"Consignment complies with Annex 7, Part A, point 74, option (b), item (i) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants have been cert under a certification scheme requiring them to be derived in direct line from material which has been maintained under appropriate conditions and has been subjected to official testing for at least Candidatus Phytoplasma 'mali' Seemüller & Schneider using appropriate indicators or equivalent methods and has been found free from that pest" OR

"Consignment complies with Annex 7, Part A, point 74, option (b), item (ii) and (iii) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants have been derived in direct line from

material which has been maintained under appropriate conditions and has been subjected, at least once within the last six complete cycles of vegetation, to official testing for at least Candidatus Phytoplasma 'mali' Seemüller & Schneider using appropriate indicators or equivalent methods and has been found free in those tests from that pest, and in either case, no symptoms of diseases caused by Candidatus Phytoplasma 'mali' Seemüller & Schneider have been observed on plants at the place of production, or on susceptible plants in its immediate vicinity, since the beginning of the last three complete cycles of vegetation" AND

"Consignment complies with Annex 7, Part A, point 72 of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that no symptoms of diseases caused by non-European viruses, viroids and phytoplasmas or *Phyllosticta solitaria* Ell. & Ev. have been observed on the plants at the place of production since the beginning of the last complete cycle of vegetation"

<u>Rosa spp.</u>

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10, 11, 17, 19 AND

"Consignment complies with Annex 7, Part A, point 72 of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants originate in an area which, in accordance with the measures specified in ISPM4, is known to be free from *Anthonomus bisignifer* Schenkling.

4.3.2 Cuttings

<u>Blechnum spp.</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6

AND

1, 2, 3, 9, 10, 11, 14 [Note: These additional declarations are conditional and may not apply.]

Cyathea spp. (C. dealbata, C. medullaris, C. smithii)

Conditions:

Import permit not required. Phytosanitary certificate, additional declaration and treatment required. Please refer to <u>MPI Approved Biosecurity Treatments</u> under section 1.8 'Nursery Stock' for treatment options.

All exports of tree ferns are subject to the Tree ferns to UK: Cuttings Only Compliance Programme and Registration on the MPI website, please refer to https://www.mpi.govt.nz/export/live-plants-laboratory-plant-products/requirementdocuments-for-exporting-plant-products/.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3, [Note: These additional declarations are conditional and may not apply.]

Dicksonia squarrosa

Conditions:

Import permit not required. Phytosanitary certificate, additional declaration and treatment required. Please refer to <u>MPI Approved Biosecurity Treatments</u> under section 1.8 'Nursery Stock' for treatment options.

All exports of tree ferns are subject to the to the Tree ferns to UK: Cuttings Only Compliance Programme and Registration on the MPI website, please refer to <u>https://www.mpi.govt.nz/export/live-plants-laboratory-plant-products/requirement-documents-for-exporting-plant-products/</u>.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3, [Note: These additional declarations are conditional and may not apply.]

<u>Epiphyllum</u> Hybrids

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14 AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

<u>Hoya polyneura</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

4.3.3 Tissue Culture

Agapanthus spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

5

AND

14 [Note: This additional declaration is conditional and may not apply.]

Beschorneria spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

5 AND

14 [Note: This additional declaration is conditional and may not apply.]

Brunnera spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 5 AND 14 [Note: This additional declaration is conditional and may not apply.] Cordyline spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 5 AND 14 [Note: This additional declaration is conditional and may not apply.] Corydalis spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 5 AND 14 [Note: This additional declaration is conditional and may not apply.] Daphne odora Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 5, 14 Dicentra spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 5 AND 14 [Note: This additional declaration is conditional and may not apply.] Gentiana spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 5, 14

Gentianella spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 5, 14 AND 15 [Note: This additional declaration is conditional and may not apply.] Geranium spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 5 AND 14 [Note: This additional declaration is conditional and may not apply.] Geum coccineum Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 5, 14 Helleborus spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 5 AND 14 [Note: This additional declaration is conditional and may not apply.] Lewisia spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 5 AND 14 [Note: This additional declaration is conditional and may not apply.] Ligularia spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration reauired. Additional declaration: 5 AND 14 [Note: This additional declaration is conditional and may not apply.]

Lomandra spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

5

AND

14 [Note: This additional declaration is conditional and may not apply.]

Magnolia spp. (M. yunnanensis)

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

5

AND

14 [Note: This additional declaration is conditional and may not apply.]

Michelia doltsopa

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

5

<u>Phormium spp.</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

5

AND

14 [Note: This additional declaration is conditional and may not apply.]

<u>Pseudopanax spp.</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

5

Tiarella spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

5

AND

14 [Note: This additional declaration is conditional and may not apply.]

Verbascum spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

5

AND

14 [Note: This additional declaration is conditional and may not apply.]

4.3.4 Tubers and corms

Paeonia lactiflora

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

5, 13

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

4.3.5 Rhizomes

<u>Hedychium gardnerianum</u>

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

5, 13

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

4.3.6 Whole Plants

<u>Acacia spp.</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration: 4, 5, 6, 10, 11, 19

4, 5, 6, 10, AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Note: Effective 14 December 2019, prohibited pending risk assessment.

<u>Acer japonicum,</u> <u>Acer palmatum,</u>

<u>Acer shirasawanum</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required. Plants of the above species must be bare rooted, free of leaves, grafted or budded, one to three-year-old plants for planting. The specification of the registered site(s) of the production must be included in the additional declarations. To meet official inspection requirements below, the size of the sample for inspection must

enable at least the detection of a 1% level of infestation with a level of confidence of 99%.

Additional declaration:

4, 5, 10, 11, 17, 19, 20

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.] AND

"Consignment complies with Annex 7, Part A, point 68A (a) of the Commission Implementing Regulation (EU) 2019/2072 as amended by the Official Controls, Plant Health, Seeds and Seed Potatoes (Amendment etc.) Regulations 2021. This declares that the plants are free from *Eotetranychus sexmaculatus* (Riley)" AND

"Consignment complies with Annex 7, Part A, point 68A (b) of the Commission Implementing Regulation (EU) 2019/2072 as amended by the Official Controls, Plant Health, Seeds and Seed Potatoes (Amendment etc.) Regulations 2021. This declares that the plants have been grown throughout their life in a place of production, which, together with the sites of production that form part of it, is registered and supervised by the national plant protection organisation of the country of origin" AND

"Consignment complies with Annex 7, Part A, point 68A (c) of the Commission Implementing Regulation (EU) 2019/2072 as amended by the Official Controls, Plant Health, Seeds and Seed Potatoes (Amendment etc.) Regulations 2021. This declares that the site of production has been found free from *Eotetranychus sexmaculatus* (Riley) during official inspections carried out at appropriate times since the beginning of the complete production cycle; in the case of suspicion of the presence of *Eotetranychus sexmaculatus* (Riley) at the site of production, appropriate treatments have been carried out to ensure the absence of the pest; a surrounding zone of 100m has been established, which is subject to specific surveys at appropriate times to detect *Eotetranychus sexmaculatus* (Riley); and where the pest has been found on any host plants, those plants have been rogued out and destroyed immediately"

AND

"Consignment complies with Annex 7, Part A, point 68A (d) of the Commission Implementing Regulation (EU) 2019/2072 as amended by the Official Controls, Plant Health, Seeds and Seed Potatoes (Amendment etc.) Regulations 2021. This declares that a system has been put in place to ensure that tools and machinery have been cleaned to be free from soil and plant debris and disinfected to be free from *Eotetranychus sexmaculatus* (Riley), before they have been introduced into each site of production"

AND

"Consignment complies with Annex 7, Part A, point 68A (e) of the Commission Implementing Regulation (EU) 2019/2072 as amended by the Official Controls, Plant Health, Seeds and Seed Potatoes (Amendment etc.) Regulations 2021. This declares that at harvest they have been cleaned and trimmed and have undergone an official phytosanitary inspection, consisting at least of a detailed visual examination, in particular of stems and branches of the plants to confirm the absence of *Eotetranychus sexmaculatus* (Riley)"

AND

"Consignment complies with Annex 7, Part A, point 68A (f) of the Commission Implementing Regulation (EU) 2019/2072 as amended by the Official Controls, Plant Health, Seeds and Seed Potatoes (Amendment etc.) Regulations 2021. This declares that immediately prior to export, the consignments have been subjected to an official inspection for the presence of *Eotetranychus sexmaculatus* (Riley), in particular of stems and branches of the plants" AND

"Consignment complies with Annex 7, Part A, point 68B (a) of the Commission Implementing Regulation (EU) 2019/2072 as amended by the Official Controls, Plant Health, Seeds and Seed Potatoes (Amendment etc.) Regulations 2021. This declares that the plants are free from *Oemona hirta* (Fabricius) and *Platypus apicalis* (White)"

AND

"Consignment complies with Annex 7, Part A, point 68B (b) of the Commission Implementing Regulation (EU) 2019/2072 as amended by the Official Controls, Plant Health, Seeds and Seed Potatoes (Amendment etc.) Regulations 2021. This declares that the plants have been grown throughout their life in a place of production, which, together with the sites of production that form part of it is registered and supervised by the national plant protection organisation of the country of origin" AND

"Consignment complies with Annex 7, Part A, point 68B (c) of the Commission Implementing Regulation (EU) 2019/2072 as amended by the Official Controls, Plant Health, Seeds and Seed Potatoes (Amendment etc.) Regulations 2021. This declares that the site of production has been found free from *Oemona hirta* (Fabricius) and *Platypus apicalis* (White) during official inspections carried out at appropriate times since the beginning of the complete production cycle; and in the case of suspicion of the presence of *Oemona hirta* (Fabricius) and *Platypus apicalis* (White) at the site of production, appropriate treatments have been carried out to ensure the absence of the pests"

AND

"Consignment complies with Annex 7, Part A, point 68B (d) of the Commission Implementing Regulation (EU) 2019/2072 as amended by the Official Controls, Plant Health, Seeds and Seed Potatoes (Amendment etc.) Regulations 2021. This declares that at harvest, they have been cleaned and have undergone an official inspection to confirm the absence of *Oemona hirta* (Fabricius) and *Platypus apicalis* (White)"

AND

"Consignment complies with Annex 7, Part A, point 68B (e) of the Commission Implementing Regulation (EU) 2019/2072 as amended by the Official Controls, Plant Health, Seeds and Seed Potatoes (Amendment etc.) Regulations 2021. This declares that immediately before export consignments have been subjected to an official inspection for the presence of *Oemona hirta* (Fabricius) and *Platypus apicalis* (White)"

ÀND

"This declares that [*Insert unique production site identifier(s)*] is a registered site(s) of production."

<u>Agapanthus spp.</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Agathis spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 10 AND 1, 2, 3,16 [Note: These additional declarations are conditional and may not apply.] Agonis flexulosa Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 9, 14 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Alectryon excelsus Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 10 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Aloe spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 10 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Apodasmia similis Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 9, 14 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Araucaria heterophylla Conditions: Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3, 16 [Note: These additional declarations are conditional and may not apply.]

Argyranthemum spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Armeria pseudarmeria

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14 AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Arthropodium spp. (A. cirrhatum)

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Asplenium spp. (Birds nest tree fern)

Grown from spores on nursery site only Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4,5,6,10 AND 1,2,3 [Note: these additional declarations are conditional and may not apply] Astelia spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Beschorneria spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Blechnum spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

<u>Boronia spp.</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Brachyscome spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Callistemon spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

<u>Campanula spp.</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate and additional declaration required. <u>Additional declaration:</u> 4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Carex spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Carpodetus serratus

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Cercis spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10, 11

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Coprosma spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Cordyline spp.

Conditions

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3, 8 [Note: These additional declarations are conditional and may not apply.]

Cornus spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration

required.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Cvathea spp. Grown from spores on nursery site only Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 10 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Dicksonia spp. Grown from spores on nursery site only Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 10 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Echeveria spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Feijoa sellowiana Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 10 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Fuchsia spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration:

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Griselinia spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 10 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Hebe spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 10, 19 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Helenium spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 9, 14 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Hemerocallis spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 9, 14, 19 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Hesperantha spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 9, 14 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Hibiscus syriacus Conditions: Import permit not required. Phytosanitary certificate and additional declaration required.
Additional declaration:

4, 5, 8, 9, 10, 11, 14, 19

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Hoheria populnea

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10, 11

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

<u>Hoya polyneura</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

<u>Iris ensata</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Knightia excelsia

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Lamium spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Leptinella spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Libertia spp. (L. ixioides)

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Lomandra spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Magnolia spp. (except M. grandiflora)

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10, 11

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Magnolia grandiflora (evergreen)

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10, 19

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Metrosideros spp.

<u>Conditions:</u> Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration:

```
4, 5, 6, 10, 19
```

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Michelia spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

<u>Olea europaea</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required. The following regulated non-quarantine pests are subject to a 0% threshold for the plants concerned (i.e. there is zero tolerance for any findings of these pests):

- Pseudomonas savastanoi pv. savastanoi (Smith) Gardan et al. [PSDMSA] (de-regulated status applies as of 2 December 2021)
- Verticillium dahliae Kleb [VERTDA]
- Pratylenchus vulnus Allen & Jensen [PRATVU]
- Xiphinema diversicaudatum (Mikoletzky) Thorne [XIPHDI]
- Arabis mosaic virus [ARMV00]
- Cherry leaf roll virus [CLRV00]
- Olive leaf yellowing associated virus [OLYAV0]
- Olive yellow mottling and decline associated virus [OYMDAV]

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.] AND

"Consignment complies with Annex 7, Part B, point 2, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that *Xylella fastidiosa* (Wells et al.) is not present in the country" AND

"Consignment complies with Annex 7, Part B, point 2, option (b) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that they have been grown in a site that is subject to annual official inspection, with sampling and testing carried out at the appropriate times on those plants for the presence of *Xylella fastidiosa* (Wells et al.) and in accordance with international standards, using a sampling scheme able to identify with 99% reliability the level of presence of infected plants of 5%, in which the absence of *Xylella fastidiosa* (Wells et al.) was confirmed"

Ophiopogon spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Pachystegia spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 9, 14 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Parahebe spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 9, 14 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Phormium spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 9, 14 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] *Phygelius* spp Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 10 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] *Pittosporum* spp. Conditions: Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration: 4, 5, 6, 10 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

<u>Pratia spp</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 9, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Pseudopanax spp. (P. ferox)

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Pseudowintera spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5,6, 10

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

Salvia spp.(except Salvia mellifera)

<u>Conditions:</u> Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 7, 9, 10, 11, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

<u>Salvia mellifera</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required. Additional declaration:

4, 5, 6, 7, 9, 10, 11, 14, 19

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

<u>Scabiosa spp.</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate and additional declaration required. <u>Additional declaration:</u> 4, 5, 6, 7, 9, 10, 11, 14 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.] Scutellaria spp.

Conditions

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 7, 9, 10, 11, 14

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

<u>Sophora spp.</u>

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required.

Additional declaration:

4, 5, 6, 10

AND

1, 2, 3, 11 [Note: These additional declarations are conditional and may not apply.] *Vaccinium* spp.

Conditions:

Import permit not required. Phytosanitary certificate and additional declaration required. The following regulated non-quarantine pests are subject to a 0% threshold for the plants concerned (i.e. there is zero tolerance for any findings of these pests):

- Agrobacterium tumefaciens (Smith & Townsend) Conn [AGRBTU]
- Exobasidium vaccinii (Fuckel) Woronin [EXOBVA]
- o Godronia cassandrae (anamorph Topospora myrtilli) Peck [GODRCA]
- Blueberry mosaic associated virus [BLMAV0]
- Blueberry red ringspot virus [BRRV00]
- Blueberry shock virus [BLSHV0]
- Candidatus Phytoplasma 'asteris' Lee et al. [PHYPAS]
- Cranberry false blossom phytoplasma [PHYPFB]

Additional declaration:

4, 5, 6, 10, 11, 19

AND

1, 2, 3 [Note: These additional declarations are conditional and may not apply.] AND

"Consignment complies with Annex 7, Part A, point 28, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020.

This declares that the plants originate in an area established by the national plant protection organisation in accordance with ISPM4 as an area that is free from *Candidatus* Phytoplasma 'solani' Quaglino et al"

OR

"Consignment complies with Annex 7, Part A, point 28, option (b) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020.

This declares that no symptoms of *Candidatus* Phytoplasma 'solani' Quaglino et al. have been observed on the plants at the place of production since the beginning of the last complete cycle of vegetation"

AND

"Consignment complies with Annex 7, Part A, point 85, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020.

This declares that the plants originate in an area, which in accordance with the measures specified in ISPM4, is known to be free from *Diaporthe vaccinii* Shear"

OR

"Consignment complies with Annex 7, Part A, point 85, option (b) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020.

This declares that no symptoms of *Diaporthe vaccinii* Shear have been observed at the production site over the last complete growing season"

AND (applies as of 2 December 2021)

"Consignment complies with Annex 7, Part A, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020.

This declares that the plants originate in a country free from *Thekopsora minima*" OR

"Consignment complies with Annex 7, Part A, option (b) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020.

This declares that the plants originate in an area* free from *Thekopsora minima*, established by the national plant protection organisation in accordance with relevant International Standards for Phytosanitary Measures"

OR

"Consignment complies with Annex 7, Part A, option (c) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020.

This declares that the plants originate in an area* free from *Thekopsora minima*, established by the national plant protection organisation in accordance with relevant International Standards for Phytosanitary Measures"

Note: Where * is indicated above, the name of the area(s) must be included in the phytosanitary certificate under the heading "Additional declaration"

<u>Wisteria spp.</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate and additional declaration required. <u>Additional declaration:</u> 4, 5, 10, 11 AND 1, 2, 3 [Note: These additional declarations are conditional and may not apply.]

4.3.7 Additional declarations for nursery stock (i.e. Plants for planting)

1. Growing medium, attached to or associated with plants, intended to sustain the vitality of the plants, with the exception of sterile medium of in-vitro plants.

Note: Where * is indicated below, the details of the treatment or the use of a systems approach must also be included on the phytosanitary certificate under the heading "Additional declaration".

Additional declaration:

"Consignment complies with Annex 7, Part A, point 1, option (a) item (i) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the growing medium at the time of their planting was free from soil and organic matter and had not been previously used for growing plants or for any other agricultural purposes and was stored and maintained under appropriate conditions to keep it free from GB quarantine pests" OR

"Consignment complies with Annex 7, Part A, point 1, option (a) item (ii) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the growing medium at the time of their planting was composed entirely of peat or fibre of *Cocos nucifera* L. and had not been previously used for growing plants or for any other agricultural purposes and was stored and maintained under appropriate conditions to keep it free from GB quarantine pests" OR

"Consignment complies with Annex 7, Part A, point 1, option (a) item (iii) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the growing medium at the time of their planting was subjected to effective fumigation or heat treatment* to ensure freedom from pests was stored and maintained under appropriate conditions to keep it free from GB quarantine pests"

OR

"Consignment complies with Annex 7, Part A, point 1, option (a) item (iv) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the growing medium at the time of their planting was subjected to an effective systems approach* to ensure freedom from pests was stored and maintained under appropriate conditions to keep it free from GB quarantine pests" AND

"Consignment complies with Annex 7, Part A, point 1, option (b) item (i) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that since planting appropriate measures have been taken to ensure that the growing medium has been kept free from GB quarantine pests, including at least:

(aa) physical isolation of the growing medium from soil and other possible sources of contamination

(bb) hygiene measures

(cc) using water free from GB quarantine pests"

OR

"Consignment complies with Annex 7, Part A, point 1, option (b) item (ii) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that since planting in the two weeks prior to export, the growing medium including, where appropriate, soil was completely removed by washing using water free from GB quarantine pests, and where replanting occurred, the growing medium used met the requirements specified in point (a) and the measures described in point (b)(i) were taken to ensure that it remains free from GB quarantine pests"

2. Plants for planting with roots, grown in open air.

Note: New Zealand is free from *Clavibacter sepedonicus* (Spieckermann & Kotthoff) Li et al.

Additional declaration:

"Consignment complies with Annex 7, Part A, point 4 of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the place of production has been established by the national plant protection organisation in accordance with ISPM10 as a place of production that is free from *Clavibacter sepedonicus* (Spieckermann & Kotthoff) Li et al. and *Synchytrium endobioticum* (Schilbersky) Percival.

3. Plants for planting with roots, grown in open air.

Additional declaration:

"Consignment complies with Annex 7, Part A, point 5 of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. The plants originate from a field known to be free from *Globodera pallida* (Stone) Behrens and *Globodera rostochiensis* (Wollenweber) Behrens"

4. Plants for planting, other than bulbs, corms, rhizomes, seeds, tubers, and plants in tissue culture.

Note: Where * is indicated below, the name of the area(s) must be included in the phytosanitary certificate under the heading "Additional declaration"

Note: New Zealand is free from *Thrips palmi* Karny.

Additional declaration:

"Consignment complies with Annex 7, Part A, point 6, option (a), item (i) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants have been grown in a nursery and originate in an area* established by the national plant protection organisation in accordance with ISPM4 as an area that is free from *Thrips palmi* Karny"

5. Plants for planting other than seeds.

Additional declaration:

"Consignment complies with Annex 7, Part A, point 7, option (a), (b), (c) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants have been grown in a nursery, are free from plant debris, flowers and fruits, and have been inspected at appropriate times and have been found prior to their export to be free from symptoms of harmful bacteria, viruses and virus-like organisms, and are free from signs or symptoms of harmful nematodes, insects, mites and fungi or have been subjected to appropriate treatment to eliminate such organisms"

6. Plants for planting, other than dormant plants, plants in tissue culture, seeds, bulbs, tubers, corms and rhizomes.

Note: The following Additional declaration applies to any third country where the following quarantine pests are known to occur: *Bemisia tabaci*, Begomoviruses, Cowpea mild mottle virus, Cucumber vein yellowing virus, Cucurbit yellow stunting disorder virus, Lettuce infectious yellows virus, Melon yellowing associated virus, Squash vein yellowing virus, Sweet potato chlorotic stunt virus, Sweet potato mild mottle virus, Tomato mild mottle virus, Tomato leaf curl New Delhi virus"

Note: New Zealand is free from Lettuce infectious yellow virus.

Additional declaration:

"Consignment complies with Annex 7, Part A, point 8, option (a) and (b), item (i) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit)

Regulations 2020. This declares that no symptoms of the relevant pests have been observed on the plants during their complete cycle of vegetation and originate in areas which, in accordance with the measures specified in ISPM4, are known to be free from *Bemisia tabaci* (Gennadius) and other vectors of the relevant pests" OR

"Consignment complies with Annex 7, Part A, point 8, option (a) and (b), item (ii) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that no symptoms of the relevant pests have been observed on the plants during their complete cycle of vegetation and that the site of production has been found free from *Bemisia tabaci* (Gennadius) and other vectors of the relevant plant pests on official inspections carried out at appropriate times to detect those pests"

OR

"Consignment complies with Annex 7, Part A, point 8, option (a) and (b), item (iii) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that no symptoms of the relevant pests have been observed on the plants during their complete cycle of vegetation and that the plants have been subjected to an effective treatment ensuring the eradication of *Bemisia tabaci* (Gennadius) and the other vectors of the relevant pests and have been found free from those pests prior to export"

7. Plants for planting, other than seeds, of Cucurbitaceae and Solanaceae.

Additional declaration:

"Consignment complies with Annex 7, Part A, point 9, option (a) item (i) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants originate in an area which, in accordance with the measures specified in ISPM4, is known to be free from Tomato leaf curl New Delhi Virus"

OR

"Consignment complies with Annex 7, Part A, point 9, option (a) item (ii) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that no symptoms of Tomato leaf curl New Delhi Virus have been observed on the plants during their complete cycle of vegetation" AND

Note: The following Additional declaration applies to any plant originating in an area where *Bemisia tabaci* (Gennadius) or other vectors of Tomato leaf curl New Delhi Virus are known to occur.

"Consignment complies with Annex 7, Part A, point 9, option (b) item (i) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the site of production has been found free from *Bemisia tabaci* (Gennadius) and other vectors of Tomato leaf curl New Delhi Virus on official inspections carried out at appropriate times to detect the pest. OR

"Consignment complies with Annex 7, Part A, point 9, option (b) item (ii) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants have been subjected to an effective treatment ensuring the eradication of *Bemisia tabaci* (Gennadius) and other vectors of Tomato leaf curl New Delhi Virus"

8. Plants for planting of Begonia L., other than seeds, tubers and corms, and

plants for planting, other than seeds, of *Ajuga* L., *Crossandra Salisbury*, *Dipladenia* A.DC., *Ficus* L., *Hibiscus* L., *Mandevilla* Lindl. and *Nerium oleander* L.

"Consignment complies with Annex 7, Part A, point 12, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants originate in an area which, in accordance with the measures specified in ISPM4, is known to be free from *Bemisia tabaci* (Gennadius)" OR

"Consignment complies with Annex 7, Part A, point 12, option (b) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that no signs of *Bemisia tabaci* (Gennadius) have been observed on plants at the place of production on official inspections carried out at least once every three weeks during the nine weeks prior to marketing" OR

"Consignment complies with Annex 7, Part A, point 12, option (c) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that in cases where *Bemisia tabaci* (Gennadius) has been found at the place of production, the plants held or produced at the place of production, have undergone an appropriate treatment to ensure freedom from *Bemisia tabaci* (Gennadius) and subsequently the place of production has been found free from *Bemisia tabaci* (Gennadius) as a consequence of the implementation of appropriate procedures aiming at eradicating *Bemisia tabaci* (Gennadius), in both official inspections carried out weekly during the three weeks prior to the movement from the place of production, the last of which was carried out immediately prior to their movement from the place of production, and in monitoring procedures throughout the period"

ÖR

"Consignment complies with Annex 7, Part A, point 12, option (d) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that in the case of plants which there is evidence from their packing or their flower development or from other means that they are intended for direct sale to final consumers not involved in professional plant production, the plants have been officially inspected and found free from *Bemisia tabaci* (Gennadius) immediately prior to their movement"

9. Plants for planting of herbaceous species, other than bulbs, corms, plants of the family Poaceae, rhizomes, seeds, tubers, and plants in tissue culture.

Note: Where * is indicated below, the name of the area(s) must be included in the phytosanitary certificate under the heading "Additional declaration"

Note: New Zealand is free from *Liriomyza sativae* Blanchard and *Nemorimyza maculosa* (Malloch).

"Consignment complies with Annex 7, Part A, point 13, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants have been grown in a nursery and originate in an area* established by the national plant protection organisation in accordance with ISPM4 as an area that is free from *Liriomyza sativae* Blanchard and *Nemorimyza maculosa* (Malloch)"

10. Trees and shrubs for planting, other than seeds and plants in tissue culture.

"Consignment complies with Annex 7, Part A, point 14, option (a), (b), (c) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants have been grown in a nursery, are free from plant debris, flowers and fruits, and have been inspected at appropriate times and prior to export and have been found to be free from:

- (i) symptoms of harmful bacteria, viruses and virus-like organisms, and
- (ii) signs or symptoms of harmful nematodes, insects, mites and fungi or have been subjected to appropriate treatment to eliminate such organisms"

11. Deciduous trees and shrubs for planting, other than seeds and plants in tissue culture.

"Consignment complies with Annex 7, Part A, point 15 of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the trees and shrubs are dormant and free from leaves"

12. Root and tubercle vegetables, other than tubers of *Solanum tuberosum* L.

"Consignment complies with Annex 7, Part A, point 16 of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the consignment or lot does not contain more than 1% by net weight of soil and growing medium"

13. Bulbs, corms, rhizomes and tubers, intended for planting, other than tubers of *Solanum tuberosum* L.

"Consignment complies with Annex 7, Part A, point 16 of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the consignment or lot does not contain more than 1% by net weight of soil and growing medium"

14. Plants of herbaceous species for planting, other than bulbs, corms, plants of the family Gramineae, rhizomes, seeds, tubers.

Note: Where * is indicated below, the name of the area(s) must be included in the phytosanitary certificate under the heading "Additional declaration"

Note: New Zealand is free from *Liriomyza huidobrensis* (Blanchard) and *Liriomyza trifolii* (Burgess).

"Consignment complies with Annex 7, Part A, point 40, option (a), of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants originate in an area* which, in accordance with the measures specified in ISPM4, is known to be free from *Liriomyza huidobrensis* (Blanchard) and *Liriomyza trifolii* (Burgess)"

15. Naturally or artificially dwarfed plants for planting other than seeds.

Note: Where * is indicated, the active ingredient, concentration and date of application of these treatments must be mentioned on the phytosanitary certificate under the

heading "disinfestation and/or disinfection treatment". Where ** is indicated, the registration number must be indicated on the phytosanitary certificate under the heading "Additional declaration".

"Consignment complies with Annex 7, Part A, point 42, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants, including those collected directly from natural habitats, have been grown, held and trained for at least two consecutive years prior to dispatch in officially registered nurseries, which are subject to an officially supervised control regime,

(b) that the plants have at least during the period referred to in point (a):

- (i) been potted, in pots which are placed on shelves at least 50 cm above ground,
- (ii) have been subjected to appropriate treatments* to ensure freedom from non- European rusts,
- (iii) have been officially inspected at least six times a year at appropriate intervals for the presence of GB quarantine pests of concern and these inspections have also been carried out on plants in the immediate vicinity of the nurseries referred to in point (a), at least by visual examination of each row in the field or nursery and by visual examination of all parts of the plant above the growing medium, using a random sample of at least 300 plants from a given genus where the number of plants of that genus is not more than 3000 plants, or 10 % of the plants if there are more than 3000 plants from that genus,
- (iv) have been found to be free, in those inspections, from the relevant GB quarantine pests of concern, infested plants have been removed and the remaining plants, where appropriate, have been effectively treated, and have been held for an appropriate period and inspected to ensure freedom from those pests,
- (v) have been planted either in an unused artificial growing medium or in a natural growing medium, which has been treated by fumigation or by appropriate heat treatment and has been found free of any GB quarantine pests, and
- (vi) have been kept under conditions which ensure that the growing medium has been maintained free from GB quarantine pests and within two weeks prior to dispatch, have been:

(aa) shaken and washed with clean water to remove the original growing medium and kept bare rooted,

(bb) shaken and washed with clean water to remove the original growing medium and replanted in growing medium which meets the conditions in point (v), or

(cc) subjected to appropriate treatments* to ensure that the growing medium is free from plant pests, and (c) that the plants have been packed in closed containers which have been officially sealed and bear the registration number** of the registered nursery"

16. Plants of Pinales, other than fruit and seeds, over 3 m in height.

"Consignment complies with Annex 7, Part A, point 44 of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants have been produced in a nursery and that they originate in a place of production which has been established by the national plant protection organisation in accordance with ISPM10 as a place of production that is free from *Scolytidae* spp. (non-European)"

17. Plants for planting, other than seeds, that have a stem or root collar diameter of 1 cm or more at their thickest point, of *Acer* spp. L., *Aesculus hippocastanum* L., *Alnus* spp. Miller, *Betula* spp. L., *Carpinus* spp., *Citrus* spp. L., *Cornus* spp., *Corylus* spp., *Cotoneaster* spp., *Crataegus* spp. L., *Fagus* spp., *Lagerstroemia* spp., *Malus* spp., *Platanus* spp. L., *Populus* spp. L., *Prunus laurocerasus* L., *Pyrus* spp., *Rosa* spp. L., *Salix* spp. L., and *Ulmus* spp. L.

"New Zealand is free from Anoplophora chinensis"

18. Live pollen of Actinidia Lindl. or plants for planting, other than seeds, of *Actinidia* Lindl., ("the specified plants")

"Consignment complies with Annex 7, Part A, point 71 option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020.

This declares that the plants have been grown throughout their life in a country where *Pseudomonas syringae* pv. actinidiae Takikawa, Serizawa, Ichikawa, Tsuyumu & Goto is known not to occur"

OR

"Consignment complies with Annex 7, Part A, point 71 option (b) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020.

This declares that the plants been grown throughout their life in a place of production which is registered and supervised by the national plant protection organisation in the country of origin and is situated in an area* established by that organisation in accordance with ISPM4 as an area that is free from *Pseudomonas syringae* pv. actinidiae Takikawa, Serizawa, Ichikawa, Tsuyumu & Goto" AND

"This declares that the specified plants have been derived directly from mother plants under conditions which comply with the requirements specified in points (a) or (b), the specified plants have been directly derived from mother plants, which were subject to prior individual testing confirming their freedom from *Pseudomonas syringae* pv. actinidiae Takikawa, Serizawa, Ichikawa, Tsuyumu & Goto, or the specified plants have been tested according to a sampling scheme that is able to confirm with 99% reliability that the level of presence of *Pseudomonas syringae* pv. actinidiae Takikawa, Serizawa, Ichikawa, Tsuyumu & Goto in the specified plants is below 0.1%" OR

"Consignment complies with Annex 7, Part A, point 71 option (c) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the plants have been produced in a place or site of production which is registered and supervised by the national plant protection organisation in the country of origin and established in accordance with the ISPM10 as a place of production that is free from *Pseudomonas syringae* pv. actinidiae Takikawa, Serizawa, Ichikawa, Tsuyumu & Goto where:

(i) they have been grown in a structure with a degree of isolation and protection from the outside environment that effectively excluded *Pseudomonas syringae* pv. actinidiae Takikawa, Serizawa, Ichikawa, Tsuyumu & Goto and have been officially inspected twice at the most appropriate times for detecting symptoms of infection during the last complete cycle of vegetation prior to their movement and found free from that pest, and (ii) the place or site of production was surrounded by a zone with a radius of at least 100 m, where: (aa) official inspections were carried out twice at the place or site and in the zone at the most appropriate times for detecting symptoms of infection during the last complete cycle of vegetation prior to their movement, and (bb) where any plants showing symptoms of infection were found during those inspections, those plants were immediately destroyed"

AND

This declares that the specified plants have been derived directly from mother plants under conditions which comply with the requirements specified in points (a) or (b), the specified plants have been directly derived from mother plants, which were subject to prior individual testing confirming their freedom from *Pseudomonas syringae* pv. actinidiae Takikawa, Serizawa, Ichikawa, Tsuyumu & Goto, or —the specified plants have been tested according to a sampling scheme that is able to confirm with 99% reliability that the level of presence of *Pseudomonas syringae* pv. actinidiae Takikawa, Serizawa, Ichikawa, Tsuyumu & Goto in the specified plants is below 0.1%" OR

"Consignment complies with Annex 7, Part A, point 71 option (d) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020.

This declares that the plants have been produced in a place of production established in accordance with ISPM10 as a place of production that is free from *Pseudomonas syringae* pv. actinidiae Takikawa, Serizawa, Ichikawa, Tsuyumu & Goto and which is surrounded:

- (i) by a zone with a radius of 500 m where: (aa) official inspections, sampling and testing have been carried out at that place of production and throughout that zone twice at the most appropriate times for detecting symptoms of infection during the last complete cycle of vegetation prior to their movement, (bb) where any plants showing symptoms of infection were found during those inspections, those plants were immediately destroyed and all specified plants in the zone were immediately destroyed or have been regularly tested at the most appropriate times and found free from that pest, and
- (ii) by a further zone lying between 500 m and 4,500 m of that place of production where: (aa) official inspections, sampling and testing have been carried out twice at the most appropriate times throughout the area for detecting symptoms of infection during the last complete cycle of vegetation prior to their movement, and (bb) where any plants showing symptoms of infection were found during those inspections, those plants were immediately destroyed and all specified plants in the further zone were immediately destroyed or have been tested according to a sampling scheme that is able to confirm with 99% reliability that the level of presence of pest in the specified plants is below 0.1%"

19. Plants for planting, other than seeds, which belong to the genera and species listed in the list of *Xylella* host plants.

Note: Refer to Appendix 1 for list of *Xylella* host plants.

Note: New Zealand is free from *Xylella fastidiosa*.

"Consignment complies with Annex 7, Part B, point 2 option (a) of the Plant

Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. *Xylella fastidiosa* (Wells et al.) is not present in the country"

20. Plants for planting, other than seeds, that have a stem diameter of 1 cm or more at their thickest point, of *Acer* spp. L., *Aesculus* spp., *Alnus* spp. Miller, *Betula* spp. L., *Carpinus* spp., *Cercidiphyllum* spp. L., *Corylus* spp., *Fagus* spp., *Fraxinus* spp L., *Koelreuteria* spp. Medikus, *Platanus* spp. L., *Populus* spp. L., *Salix* spp. L., *Tilia* spp. and *Ulmus* spp. L.

"New Zealand is free from Anoplophora glabripennis"

<u>All other plants for planting</u> <u>Conditions:</u> Requirements not specified. Contact <u>PlantExports@mpi.govt.nz</u>.

4.4 Seed, Grain and Nuts for sowing

Conditions:

Refer Section 3.8.

<u>Acacia baileyana purpurea</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Acacia dealbata</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Achillea millefolium</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Allium ascalonicum</u> Conditions:

Import permit not required. Phytosanitary certificate required.

<u>Allium cepa</u> <u>Onion</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate and Additional declaration required.

Additional declaration:

"Consignment complies with Annex 5, Part E, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the crop has been visually inspected at least once at an appropriate time to detect *Ditylenchus dipsaci* (Kuehn) Filipjev since the beginning of the last complete cycle of vegetation and no symptoms of that pest have been observed" OR

<u>Shallot</u>

Silver wattle

"Consignment complies with Annex 5, Part E, option (b) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the harvested seeds have been found to be free of *Ditylenchus dipsaci* (Kuehn) Filipjev after laboratory tests on a representative sample" OR

"Consignment complies with Annex 5, Part E, option (c) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the planting material has been subjected to an appropriate chemical or physical treatment against *Ditylenchus dipsaci* (Kuehn) Filipjev and the seeds have been found to be free of that pest after laboratory tests on a representative sample"

<u>Allium fistulosum</u>

Spring onion

Conditions:

Import permit not required. Phytosanitary certificate required.

<u>Allium porrum</u>

Leek

Conditions:

Import permit not required. Phytosanitary certificate and Additional declaration required.

Additional declaration:

"Consignment complies with Annex 5, Part E, option (a) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the crop has been visually inspected at least once at an appropriate time to detect *Ditylenchus dipsaci* (Kuehn) Filipjev since the beginning of the last complete cycle of vegetation and no symptoms of that pest have been observed" OR

"Consignment complies with Annex 5, Part E, option (b) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the harvested seeds have been found to be free of *Ditylenchus dipsaci* (Kuehn) Filipjev after laboratory tests on a representative sample" OR

"Consignment complies with Annex 5, Part E, option (c) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the planting material has been subjected to an appropriate chemical or physical treatment against *Ditylenchus dipsaci* (Kuehn) Filipjev and the seeds have been found to be free of that pest after laboratory tests on a representative sample"

Allium schoenoprasum

<u>Chives</u>

Conditions:

Import permit not required. Phytosanitary certificate required.

Asparagus officinalis

Asparagus

<u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Avena sativa</u>		<u>Oats</u>	
Conditions:			
Import permit not requ	uired. Phytosanitary cert	ificate required.	
ICPR GREAT BRITAIN	Status: APPROVED	19 September 2023	

<u>Barbarea verna</u>

Conditions:

Import permit not required. Phytosanitary certificate required.

<u>Beta vulgaris</u>

Conditions:

<u>Beet</u>

Import permit not required. Phytosanitary certificate required.

<u>Brassicaceae (except Brassica napus, Brassica rapa)</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Brassica napus</u>

Conditions:

Import permit not required. Phytosanitary certificate required.

No more than 10 *Sclerotinia sclerotiorum* or fragments of *Sclerotinia sclerotiorum* found in a laboratory examination of a representative sample of each seed lot of the specified size (100g), if any.

<u>Brassica rapa</u>

Conditions:

Import permit not required. Phytosanitary certificate required.

No more than 5 *Sclerotinia sclerotiorum* or fragments of *Sclerotinia sclerotiorum* found in a laboratory examination of a representative sample of each seed lot of the specified size (70g), if any.

Capsicum spp.

Capsicum/Chilli peppers

Conditions:

Import permit not required. Phytosanitary certificate and Additional declaration required.

Additional declaration:

Great Britain have a threshold for regulated non-quarantine pests (RNQPs). For a full list of RNQPs with categories and thresholds can be referred to in Annex 4 & Annex 5 of The Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020.

Specifically, the thresholds for *Capsicum* spp. seeds are shown in the table below:

Part E, Annex 4 and 5: RNQPs concerning *Capsicum annuum* L. vegetable seed

Fungi and oomycetes	Fungi and oomycetes				
RNQPs or symptoms			Requirements		
caused by RNQPs	plants	for the			
	(seeds)	vegetable			
	(genus or	seed			
	species)	concerned			

Xanthomonas euvesicatoria Jones et al. [XANTEU]	Capsicum annuum L.	0%	Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that: (a) the seeds originate in areas known to free from <i>Xanthomonas</i> <i>euvesicatoria</i> Jones et al., AND (b) no symptoms of disease caused by <i>Xanthomonas euvesicatoria</i> Jones et al. have been observed on visual inspections at appropriate times to detect the pest during the complete cycle of vegetation of the plants at the site of production, OR (c)the seeds have been subjected to official testing for <i>Xanthomonas</i> <i>euvesicatoria</i> Jones et al. on a representative sample using appropriate methods (whether or not following an appropriate treatment) and have been found in those tests to be free from that pest.
Xanthomonas gardneri (ex Šutič 1957) Jones <i>et al.</i> [XANTGA]	Capsicum annuum L.	0%	Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that: a) the seeds originate in areas known to be free from <i>Xanthomonas</i> gardneri (ex Šutič) Jones et al., AND (b) no symptoms of disease caused by <i>Xanthomonas</i> gardneri (ex Šutič) Jones et al. have been observed on visual inspections at appropriate times to detect the pest during the complete cycle of vegetation of the plants at the site of production, OR (c) the seeds have been subjected to official testing for <i>Xanthomonas</i> gardneri (ex Šutič) Jones et al. on a representative sample using appropriate methods (whether or not following an appropriate treatment) and have been found in those tests to be free from that pest.

Xanthomonas perforans Jones et al. [XANTPF]	Capsicum annuum L.	0%	Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that:
			 (a) the seeds originate in areas known to be free from <i>Xanthomonas perforans</i> Jones et al., AND (b) no symptoms of disease caused by <i>Xanthomonas perforans</i> Jones et al. have been observed on visual inspections at appropriate times during the complete cycle of vegetation of the plants at the site of production, OR
			(c) the seeds have been subjected to official testing for <i>Xanthomonas</i> <i>perforans</i> Jones et al. on a representative sample using appropriate methods (whether or not following an appropriate treatment) and have been found in those tests to be free from that pest.
<i>Xanthomonas vesicatoria</i> (ex Doidge) Vauterin <i>et al.</i> [XANTVE]	Capsicum annuum L.	0%	Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that:
			 (a) the seeds originate in areas known to be free from <i>Xanthomonas vesicatoria</i> (ex Doidge) Vauterin et al., AND (b) no symptoms of disease caused by <i>Xanthomonas vesicatoria</i> (ex Doidge) Vauterin et al. have been observed on visual inspections at appropriate times during the complete cycle of vegetation of the plants at the site of production, OR
			(c)the seeds have been subjected to official testing for <i>Xanthomonas</i> <i>vesicatoria</i> (ex Doidge) Vauterin et al. on a representative sample using appropriate methods (whether or not following an appropriate treatment) and have been found in those tests to be free from that pest.
Viruses, viroids, virus – ICPR GREAT BRITAIN	Status: APPROV		Iasmas 9 September 2023 Page 56 of 87

Potato spindle viroid [PSTVD0]	tuber	Capsicum annuum L.	0%	Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that:
				 (a) the seeds originate in areas where Potato spindle tuber viroid is not known to occur, AND (b) no symptoms of diseases caused by Potato spindle tuber viroid have been observed on the plants at the place of production during their complete cycle of vegetation, OR (c) the seeds have been subjected to official testing for Potato spindle tuber viroid, on a representative sample using appropriate methods and have been found in those tests to be free from that pest.

AND

"Consignment complies with Annex 7, Part B, Point 6, option b, item (i) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the mother plants of seeds have been produced in a production site* where Tomato brown rugose fruit virus is known not to occur on the basis of official inspections carried out at the appropriate time to detect that pest." OR

"Consignment complies with Annex 7, Part B, Point 6, option b, item (i) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares the seeds were harvested prior to 15 August 2020 and mother plants are no longer available for inspection"

AND

"Consignment complies with Annex 7, Part B, Point 6, option b, item (ii) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the seeds or their mother plants have undergone official sampling and testing for tomato brown rugose fruit virus and have been found, according to those tests, to be free from that pest.

Note: The above additional declaration which specifies "prior to 15 August 2020" is a temporary alternative until 31 December 2023.

Note: The above additional declaration which specifies "undergone official sampling and testing" is to be in accordance with the sampling schemes and testing methods listed in Appendix 3.

Note: The name of the site(s) of production must be included in the phytosanitary certificate under the heading "Additional declaration".

<u>Cicer arientum</u> <u>Conditions:</u>	<u>Chickpea</u>
Import permit not required. Phytosanitary certificate require <u>Cichorium endivia</u> <u>Conditions:</u>	α.
Import permit not required. Phytosanitary certificate require	d.
<u>Cichorium intybus</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate require	<u>Chicory</u> d.
<u>Coriandrum sativum</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate require	<u>Coriander</u> d.
<u>Crambe abyssinica</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate require	d.
	bage, Cauliflower,Sprouts,
<u>Conditions:</u> Import permit not required. Phytosanitary certificate require	<u>ard, Turnip, Broccoli</u> d.
<u>Cucurbita maxima</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate require	d.
<u>Cucurbita moschata</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate require	d.
<u>Cucurbita pepo</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate require	d.
<u>Cupressus macrocarpa</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate require	d.
<u>Cynosurus cristatus</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate require	d.
<u>Daucus carota</u> <u>Conditions:</u>	Carrot
Import permit not required Phytosanitary certificate require	h

Import permit not required. Phytosanitary certificate required.ICPR GREAT BRITAINStatus: APPROVED19 September 2023

<u>Eucalyptus spp.</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.	
<u>Festulolium spp.</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.	
<u>Gramineae</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.	Grasses and Cereals
<u>Hordeum vulgare</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.	
<u>Lactuca sativa</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.	Lettuce
<u>Lathyrus odoratus</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.	Sweet pea
<u>Lolium multiflorum</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.	
<u>Lolium perenne</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.	
<u>Lolium spp.</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.	
<u>Mentha spicata</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.	
<u>Ocimum basilicum</u> Conditions:	

Import permit not required. Phytosanitary certificate required.

<u>Eruca sativa</u> Conditions:

Conditions:Import permit not required. Phytosanitary certificate required.ICPR GREAT BRITAINStatus: APPROVED19 September 2023

<u>Origanum hirtum</u>

Conditions:

Import permit not required. Phytosanitary certificate required.

<u>Origanum vulgare</u>

Conditions:

Import permit not required. Phytosanitary certificate required.

Pastinaca sativa

Conditions:

Import permit not required. Phytosanitary certificate required.

Petroselinum crispum

<u>Conditions:</u> Import permit not required. Phytosanitary certificate required. <u>Petroselinum spp.</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Pisum elatius</u>

Conditions:

Import permit not required. Phytosanitary certificate required.

<u>Pisum sativum</u>

Pea

Conditions:

Import permit not required. Phytosanitary certificate and Additional declaration required.

Additional declaration:

"Consignment complies with Annex 5, Part E of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that a representative sample of the seed has been subject to visual inspection at the most appropriate time to detect *Bruchus pisorum* (L.), which may be following an appropriate treatment, and the seed has been found to be free from that pest"

Plantago lanceolata

Conditions: Import permit not required. Phytosanitary certificate required.

Raphanobrassica spp.

Conditions:

Import permit not required. Phytosanitary certificate required.

<u>Raphanus sativus</u>

Conditions:

Import permit not required. Phytosanitary certificate required.

Raspberry, Blackberry

<u>Rubus spp.</u>

Conditions:

Import permit not required. Phytosanitary certificate required.

Solanum lycopersicum

<u>Tomato</u>

Conditions:

Import permit not required. Phytosanitary certificate and Additional declaration required.

Additional declaration:

Great Britain have a threshold for regulated non-quarantine pests (RNQPs). A full list of RNQPs with categories and thresholds can be referred to in Annex 4 & Annex 5 of The Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020.

Specifically, the thresholds for *Solanum lycopersicum* seeds are shown in the table below:

Part E, Annex 4 and 5: RNQPs concerning *Solanum lycopersicum* vegetable seed

Fungi and oomycetes			
RNQPs or symptoms caused by RNQPs	Plants for plants (seeds) (genus or species)	Thresholds for the vegetable seed concerned	Requirements
Candidatus Liberibacter 'solanacearum' Liefting et al. [LIBEPS]	Solanum lycopersicum L.	0%	-
Clavibacter michiganensis ssp. michiganensis (Smith) Davis et al. [CORBMI]	Solanum lycopersicum	0%	Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that: (a)the seeds have been obtained by means of an appropriate acid extraction method or an equivalent method, AND (i)the seeds originate in areas known to be free from <i>Clavibacter michiganensis</i> ssp. <i>michiganensis</i> (Smith) Davis et al., (ii)no symptoms of disease caused by <i>Clavibacter</i> <i>michiganensis</i> (Smith) Davis et al. have been observed on visual inspections at appropriate times

Xanthomonas euvesicatoria Jones et al. [XANTEU]Solanum lycopersicum L.0%Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that:(a) the seeds have been obtained by an appropriate acid extraction, and originate in areas known to free from Xanthomonas euvesicatoria Jones et al. () no symptoms of disease caused by Xanthomonas euvesicatoria Jones et al. have been observed on visual inspections at appropriate times to detect the pest during the complete cycle of vegetation of the plants at the site of production, OR (ii) the seeds have been subjected to official testing for Xanthomonas euvesicatoria Jones et al. [XANTGA]Xanthomonas gardneri (ex Šutič 1957) Jones et al. [XANTGA]Solanum L.0%Xanthomonas gardneri (ex Šutič 1957) L.Solanum L.0%Kanthomonas gardneri (ex Šutič 1957) L.0%Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This				to detect the pest during the complete cycle of vegetation of the plants at the site of production, OR (iii)the seeds have been subjected to official testing for <i>Clavibacter michiganensis</i> ssp. <i>michiganensis</i> (Smith) Davis et al. on a representative sample using appropriate methods and have been found in those tests to be free from that pest.
Xanthomonas gardneri (ex Šutič 1957) Jones et al. [XANTGA]Solanum lycopersicum L.0%Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU	euvesicatoria Jones et	lycopersicum	0%	Annex 5, Part E, of the PlantHealth(PhytosanitaryConditions) (Amendment) (EUExit) Regulations 2020. This
gardneri (ex Šutič 1957)lycopersicumAnnex 5, Part E, of the PlantJones et al. [XANTGA]L.Health(Phytosanitary Conditions) (Amendment) (EU				obtained by an appropriate acid extraction, and originate in areas known to free from Xanthomonas euvesicatoria Jones et al., AND (b) either: (i) no symptoms of disease caused by Xanthomonas euvesicatoria Jones et al. have been observed on visual inspections at appropriate times to detect the pest during the complete cycle of vegetation of the plants at the site of production, OR (ii) the seeds have been subjected to official testing for Xanthomonas euvesicatoria Jones et al. on a representative sample using appropriate methods (whether or not following an appropriate treatment) and have been found in those tests to be free from that pest.
declares that:	gardneri (ex Šutič 1957)	lycopersicum		Annex 5, Part E, of the PlantHealth(PhytosanitaryConditions) (Amendment) (EUExit) Regulations 2020. This

Xanthomonas perforans Jones et al. [XANTPF]	Solanum Iycopersicum L.	0%	 (a)the seeds have been obtained by an appropriate acid extraction and originate in areas known to be free from <i>Xanthomonas gardneri</i> (ex Šutič) Jones et al., AND (b) either: (i) no symptoms of disease caused by <i>Xanthomonas gardneri</i> (ex Šutič) Jones et al. have been observed on visual inspections at appropriate times during the complete cycle of vegetation of the plants at the site of production, OR (ii) the seeds have been subjected to official testing for <i>Xanthomonas gardneri</i> (ex Šutič) Jones et al. on a representative sample and using appropriate methods (whether or not following an appropriate treatment) and have been found in those tests to be free from that pest. Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This
ICPR GREAT BRITAIN	Status: APPROVED		declares that: (a) the seeds have been obtained by an appropriate acid extraction and originate in areas known to be free from <i>Xanthomonas perforans</i> Jones et al., OR (i) no symptoms of disease caused by <i>Xanthomonas</i> <i>perforans</i> Jones et al. have been observed on visual inspections at appropriate times during the complete cycle of vegetation of the plants at the site of production, OR (ii) the seeds have been subjected to official testing for <i>Xanthomonas perforans</i> Jones et al. on a representative sample using appropriate methods (whether or not following an

			appropriate treatment) and have been found in those tests to be free from that pest
Xanthomonas vesicatoria (ex Doidge) Vauterin et al. [XANTVE]	Solanum lycopersicum L.	0%	free from that pest. Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that: (a) the seeds have been obtained by an appropriate acid extraction and originate in areas known to be free from Xanthomonas vesicatoria (ex Doidge) Vauterin et al., (b) no symptoms of disease caused by Xanthomonas vesicatoria (ex Doidge) Vauterin et al. have been observed on visual inspections at appropriate times during the complete cycle
Viruses, viroids, viru	ıs – like dis	seases and	of vegetation of the plants at the site of production, OR (c) the seeds have been subjected to official testing for <i>Xanthomonas vesicatoria</i> (ex Doidge) Vauterin et al. on a representative sample using appropriate methods (whether or not following an appropriate treatment) and have been found in those tests to be free from that pest.
phytoplasmas Pepino mosaic virus [PEPMV0]	Solanum lycopersicum L.	0%	Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that:
			 (a) the seeds have been obtained by means of an appropriate acid extraction method or an equivalent method, AND (i) the seeds originate in areas where Pepino mosaic virus is known not to occur, (ii) no symptoms of diseases caused by Pepino mosaic virus

				have been observed on the plants at the place of production during their complete cycle of vegetation, OR (iii) the seeds have been subjected to official testing for Pepino mosaic virus, on a representative sample using appropriate methods, and have been found in those tests to be free from that pest.
Potato spindle t viroid [PSTVD0]	uber	Solanum lycopersicum L.	0%	Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that: (a) the seeds originate in areas where Potato spindle tuber viroid is not known to occur, (b) no symptoms of diseases caused by Potato spindle tuber viroid have been observed on the plants at the place of production during their complete cycle of vegetation, OR (c) the seeds have been subjected to official testing for Potato spindle tuber viroid, on a representative sample using appropriate methods and have been found in those tests to be free from that pest.
Tomato apical s viroid [TASVD0]	stunt	Solanum lycopersicum L.	0%	Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that:(a) the seeds originate in areas where Tomato apical stunt viroid is not known to occur, (b) no symptoms of diseases caused by Tomato apical stunt viroid have been observed on the plants at the place of production during their complete cycle of vegetation, OR (c) the seeds have been subjected to official testing for Tomato apical stunt viroid on a

			representative sample using appropriate methods and have been found in those tests to be free from that pest.
Tomato chlorotic dwarf viroid [TCDVD0]	Solanum lycopersicum L.	0%	Consignment complies with Annex 5, Part E, of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that: (a) the seeds originate in areas where Tomato chlorotic dwarf viroid is not known to occur, (b) no symptoms of diseases caused by Tomato chlorotic dwarf viroid have been observed on the plants at the place of production during their complete cycle of vegetation, OR (c) the seeds have been subjected to official testing for Tomato chlorotic dwarf viroid on a representative sample using appropriate methods and have been found in those tests to be free from that pest.

AND

"Consignment complies with Annex 7, Part B, Point 6, option b, item (i) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the mother plants of seeds have been produced in a production site* where Tomato brown rugose fruit virus is known not to occur on the basis of official inspections carried out at the appropriate time to detect that pest." OR

"Consignment complies with Annex 7, Part B, Point 6, option b, item (i) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares the seeds were harvested prior to 15 August 2020 and mother plants are no longer available for inspection"

AND

"Consignment complies with Annex 7, Part B, Point 6, option b, item (ii) of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that the seeds or their mother plants have undergone official sampling and testing for tomato brown rugose fruit virus and have been found, according to those tests, to be free from that pest.

Note: The above additional declaration which specifies "prior to 15 August 2020" is a temporary alternative until 31 December 2023.

Note: The above additional declaration which specifies "undergone official sampling and testing" is to be in accordance with the sampling schemes and testing methods

listed in Appendix 3.

Note: The name of the site(s) of production must be included in the phytosanitary certificate under the heading "Additional declaration".

<u>Spinacia oleracea</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Swinglea spp.</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Thymus vulgaris</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Triticum aestivum</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Triticosecale spp.</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

Trifolium spp.CloverConditions:Import permit not required. Phytosanitary certificate required.

<u>Valerianella locusta</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Vicia faba</u> Import permit not required. Phytosanitary certificate required. <u>Additional declaration:</u>

"Consignment complies with Annex 5, Part E of the Plant Health (Phytosanitary Conditions) (Amendment) (EU Exit) Regulations 2020. This declares that a representative sample of the seed has been subject to visual inspection at the most appropriate time to detect *Bruchus rufimanus* L., which may be following an appropriate treatment, and the seed has been found to be free from that pest"

<u>All other seed</u> <u>Conditions:</u> Requirements not specified. Contact <u>PlantExports@mpi.govt.nz</u>.

4.4.2 Seeds, Grains and Nuts for Processing

<u>Limnanthes alba</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Limnanthes spp.</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Oenothera biennis</u> <u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

<u>Oenothera spp.</u>

<u>Conditions:</u> Import permit not required. Phytosanitary certificate required.

4.5 Growing media

Growing media

PROHBITED, unless composed entirely of peat or fibre of *Cocos nucifera,* previously not used for growing of plants of any agricultural purposes. **Note:** Peat includes *Sphagnum* moss.

<u>Sphagnum spp.</u>

Conditions:

Import permit not required. Phytosanitary certificate required.

4.6 Miscellaneous

Pelleted Hops (Humulus lupulus)

Conditions:

Import permit not required. Phytosanitary certificate not required. Attestation certificate of equivalence required.

Note: Hops imported into GB until 1 February 2025, are required to be accompanied by an EU Attestation of Equivalence.

For further information please see:

Commission Regulation (EC) No 1295/2008 of 18 December 2008 on the importation of hops from third countries (Codified version)

Note: Annex II of this regulation is a template of an EU Attestation certificate of equivalence.

Hop powder (Humulus lupulus)

Conditions:

Import permit not required. Phytosanitary certificate not required. Attestation certificate of equivalence required.

Note: Hops imported into GB until 1 February 2025, are required to be accompanied by an EU Attestation of Equivalence.

For further information please see:

<u>Commission Regulation (EC) No 1295/2008 of 18 December 2008 on the importation of hops from third countries (Codified version)</u>

Note: Annex II of this regulation is a template of an EU Attestation certificate of equivalence.

Appendix 1: List of Xylella host plants

<i>Acacia</i> spp.	Acer spp.	Albizia julibrissin	Alnus rhombifolia	Ambrosia spp.
Ampelopsis arborea	Ampelopsis brevipedunculata	Ampelopsis cordata	Anthyllis hermanniae	Artemisia spp.
Asparagus acutifolius	Baccharis spp.	Brassica spp.	Calicotome spinosa	Calicotome villosa
Callicarpa americana	Callistemon citrinus	Calluna vulgaris	Carya spp.	Catharanthus spp.
Celtis occidentalis	Cercis canadensis	Cercis occidentalis	Cercis siliquastrum	Chamaecrista fasciculata
Chamaesyce canescens	Chenopodium album	Chionanthus spp.	Chitalpa tashkentensis	Cistus spp.
Citrus spp.	Clematis cirrhosa	Coelorachis cylindrica	Coffea spp.	Conium maculatum
Convolvulus cneorum	Coprosma repens	Coronilla glauca	Coronilla valentina	Cyperus eragrostis
Cytisus spp.	Digitaria spp.	Dimorphoteca spp.	Diospyros kaki	Diplocyclos palmatus
Dodonaea viscosa	Elaeagnus angustifolia	Encelia farinosa	Eremophila maculata	Erigeron spp.
Erodium moschatum	Erysimum spp.	Euphorbia chamaesyce	Euphorbia terracina	Euryops chrysanthemoides
Euryops pectinatus	Fagus crenata	Fallopia japonica	Fatsia japonica	Ficus carica
Frangula alnus	Fraxinus spp.	Genista spp.	Ginkgo biloba	Gleditsia triacanthos
Grevillea juniperina	Hebe spp.	Helianthus spp.	Helichrysum spp.	Heliotropium europaeum
Hemerocallis spp.	Hevea brasiliensis	Hibiscus spp.	Humulus scandens	llex aquifolium
llex vomitoria	Iva annua	Jacaranda mimosifolia	Juglans spp.	Juniperus ashei
Koelreuteria bipinnata	Lagerstroemia spp.	Laurus nobilis	Lavandula spp.	Ligustrum lucidum

ICPR GREAT BRITAIN

Liquidambar styraciflua	Lonicera japonica	Lupinus spp.	Magnolia grandiflora	Mallotus paniculatus
Malva parviflora	Medicago arborea	Medicago sativa	Metrosideros spp.	Mimosa spp.
Modiola caroliniana	Morus spp.	Myoporum insulare	Myrtus communis	Nandina domestica
Neptunia lutea	Nerium oleander	Olea spp.	Osteospermum ecklonis	Osteospermum fruticosum
Parthenocissus quinquefolia	Paspalum dilatatum	Pelargonium spp.	Persea americana	Phagnalon saxatile
Phillyrea angustifolia	Phillyrea latifolia	Phlomis fruticosa	Phoenix reclinata	Phoenix roebelenii
Pinus taeda	Pistacia vera	Plantago lanceolata	Platanus spp.	Pluchea odorata
Polygala myrtifolia	Polygala x grandiflora	Prunus spp.	Pterospartum tridentatum	Pyrus spp.
Quercus spp.	Ratibida columnifera	Rhamnus alaternus	Rhus spp.	Robinia pseudoacacia
Rosa spp.	Rosmarinus officinalis	Rubus spp.	Salvia mellifera	Sambucus spp.
Santolina chamaecyparissu	Sapindus saponaria	Sassafras spp.	Setaria magna	Solidago fistulosa
Solidago virgaurea	Sorghum halepense	Spartium spp.	Stewartia pseudocamellia	Strelitzia reginae
Streptocarpus spp.	Symphyotrichum divaricatum	Teucrium capitatum	Trifolium repens	Ulex spp.
<i>Ulmu</i> s spp.	Vaccinium spp.	Vinca spp.	Vitis spp.	Westringia fruticosa
Westringia glabra	Xanthium strumarium			

Appendix 2: Quarantine Pest List

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
Arachnid	Acarida (mites)	Eriophyidae	Aculops fuchsiae
			Phyllocoptes fructiphilus
		Tetranychidae	Eotetranychus lewisi
			Eutetranychus orientalis
			Eotetranychus sexmaculatus
			Oligonychus perditus
			Tetranychus evansi
Bacterial disease and phytoplasma	Acholeplasmatales	Acholeplasmataceae	Apricot chlorotic leafroll phytoplasma syn. Phytoplasma prunorum
			<i>Candidatus</i> Phytoplasma 'aurantifolia' syn. Lime witches' broom phytoplasma syn. Phytoplasma aurantifolia
			Candidatus Phytoplasma australiense
			Candidatus Phytoplasma 'mali' syn. Citrus witches' broom phytoplasma
			Candidatus Phytoplasma phoenicium
			Candidatus Phytoplasma 'pruni' syn. Peach X disease
			phytoplasma syn. Phytoplasma pruni <i>Candidatus</i> Phytoplasma 'solani' syn. Potato stolbur
			phytoplasma syn. Phytoplasma solani
			<i>Candidatus</i> Phytoplasma 'ulmi' syn. Phytoplasma ulmi syn. Elm phloem necrosis
			Coconut lethal yellowing phytoplasma syn. Palm lethal yellowing phytoplasma
			Grapevine flavescence dorée phytoplasma
			Peach rosette phytoplasma
Bacterial disease			Peach yellows phytoplasma
and phytoplasma			Strawberry witches' broom phytoplasma
	Actinomycetales	Microbacteriaceae	Clavibacter michiganensis subsp. insidiosus
			Clavibacter michiganensis subsp. Michiganensis syn. Corynebacterium michiganense
ICPR GREAT BRITAIN	Status: APPROVED	19 September 2023	Page 72 of 87
PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
-------------------	-------------------	--------------------	---
			Clavibacter sepedonicus
			Curtobacterium flaccumfaciens pv. flaccumfaciens
	Burkholderiales	Burkholderiaceae	Pseudomonas caryophylli syn. Burkholderia caryophylli
			Ralstonia pseudosolanacearum
			Ralstonia syzygii subsp. celebesensis
			Ralstonia syzygii subsp. indonesiensis
			Ralstonia solanacearum
		Unassigned	Xylophilus ampelinus
	Enterobacteriales	Enterobacteriaceae	Erwinia amylovora
			Erwinia chrysanthemi pv. Dianthicola syn. Dickeya dianthicola
			Erwinia stewartii syn. Pantoea stewartii
			Pantoea stewartii subsp. stewartii
	Entomoplasmatales	Spiroplasmataceae	Spiroplasma citri
	Lysobacterales	Lysobacteraceae	Citrus variegated chlorosis agent
			Xanthomonas arboricola pv. pruni
			Xanthomonas axonopodis pv. vesicatoria
			Xanthomonas campestris pv. oryzae
			Xanthomonas campestris pv. phaseoli
			Xanthomonas citri pv. aurantifolii
Bacterial disease			Xanthomonas citri pv. citri
and phytoplasma			Xanthomonas fragariae
			Xanthomonas oryzae pv. oryzicola
			Xylella fastidiosa
	Pseudomonadales	Pseudomonadaceae	Pseudomonas syringae pv. actinidiae
			Pseudomonas syringae pv. persicae
	Rhizobiales	Phyllobacteriaceae	Candidatus liberibacter spp.
	Unassigned	Unassigned	Peach phony rickettsia
			Non-European phytoplasmas of <i>Cydonia</i> Mill., <i>Fragaria</i> L., <i>Malus</i> Mill., <i>Prunus</i> L., <i>Pyrus</i> L., <i>Ribes</i> L., <i>Rubus</i> L. and <i>Vitis</i> L.

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
Chromista	Peronosporales	Peronosporaceae	Phytophthora fragariae var. fragariae
			Phytophthora kernoviae
			Phytophthora ramorum
			Plasmopara halstedii
Fungal disease	Botryosphaeriales	Botryosphaeriaceae	Botryosphaeria kuwatsukai
(including			Botryosphaeria laricina
oomycetes)		Phyllostictaceae	Phyllosticta citricarpa
			Phyllosticta solitaria
	Cantharellales	Hydnaceae	Trechispora brinkmannii syn. Sistotrema brinkmannii
	Capnodiales	Mycosphaerellaceae	Cercospora angolensis syn. Pseudocercospora angolensis
			Dothistroma pini
			Lecanosticta acicula syn. Scirrhia acicola
			Mycosphaerella larici-leptolepis
			Sphaerulina musiva syn. Mycosphaerella populorum
			Pseudocercospora pini-densiflorae syn. Cercoseptoria pini- densiflorae syn. Mycosphaerella gibsonii
Fungal disease			Scirrhia pini syn. Dothistroma septosporum
(including oomycetes)			Septoria malagutii syn. Septoria lycopersici var. malagutii
UUIIIyCeles)	Chaetothyriomycetidae	Chaetothyriales	Phialophora cinerescens
	Chytridiales	Synchytriaceae	Synchytrium endobioticum
	Diaporthales	Cryphonectriaceae	Cryphonectria parasitica syn. Endothia parasitica
		Diaporthaceae	Diaporthe vaccinii
		Valsaceae	Anisogramma anomala
			Stegophora ulmea
	Helotiales	Dermateaceae	Atropellis apiculata
			Atropellis pinicola
			Atropellis piniphila
			Atropellis tingens
		Helotiaceae	Chalara fraxinea syn. Hymenoscyphus fraxineus

Page **74** of **87**

19 September 2023

ICPR GREAT BRITAIN

Status: APPROVED

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
			Hymenoscyphus pseudoalbidus syn. Hymenoscyphus
			fraxineus
	Hymenochaetales	Hymenochaetaceae	Coniferiporia sulphurascens
			Coniferiporia weirii
	Hypocreales	Bionectriaceae	Geosmithia morbida
		Nectriaceae	Fusarium euwallaceae
		Hypocreaceae	Verticillium albo-atrum
			Verticillium dahliae
		Nectriaceae	Fusarium oxysporum f. sp. albedinis
			Fusarium circinatum syn. Gibberella circinata
Fungal disease	Microascales	Ceratocystidaceae	Bretziella fagacearum
(including			Ceratocystis platani
oomycetes)			Davidsoniella virescens syn. Ceratocystis virescens
	Myriangiales	Elsinoaceae	Elsinoe spp.
	Pezizales	Rhizinaceae	Phymatotrichopsis omnivora
	Pleosporales	Didymellaceae	Didymella ligulicola syn. Stagonosporopsis chrysanthemi
			Stagonosporopsis andigena syn. Phoma andina
		Leptosphaeriaceae	Phoma tracheiphila syn. Plenodomus tracheiphilus
		Pleosporaceae	Alternaria alternata
			Alternaria mali
		Venturiaceae	Apiosporina morbosa
			Venturia nashicola
	Pucciniales	Coleosporiaceae	Chrysomyxa arctostaphyli
		Cronartiaceae	Cronartium spp. (except Cronartium gentianeum, Cronartium
			pini and Cronartium ribicola)
			Endocronartium spp.
		Melampsoraceae	Melampsora farlowii
			Melampsora medusae
			Melampsora medusae f. sp. tremuloidis

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
Fungal disease (including oomycetes)		Pucciniaceae	Gymnosporangium spp. (except Gymnosporangium amelanchieris, Gymnosporangium atlanticum, Gymnosporangium clavariiforme, Gymnosporangium confusum, Gymnosporangium cornutum, Gymnosporangium fusisporum, Gymnosporangium gaeumannii, Gymnosporangium gracile, Gymnosporangium minus, Gymnosporangium orientale, Gymnosporangium sabinae, Gymnosporangium torminali-juniperini and Gymnosporangium tremelloides)
			Puccinia pittieriana
		Pucciniastraceae	<i>Thekopsora minima</i> (Note: quarantine pest until 25 November 2022)
	Russulales	Bondarzewiaceae	Heterobasidion irregulare
	Tilletiales	Tilletiaceae	Tilletia indica syn. Neovossia indica
	Ustilaginales	Glomosporiaceae	Thecaphora solani syn. Angiosorus solani
	Xylariales	Xylariaceae	Hypoxylon mammatum syn. Entoleuca mammata
Insect	Coleoptera (beetles,	Brentidae	Arrhenodes minutus
	weevils)	Buprestidae	Agrilus anxius
			Agrilus bilineatus
			Agrilus fleischeri
			Agrilus plannipennis
		Cerambycidae	Anoplophora chinensis
			Anoplophora glabripennis
			Neocerambyx raddei syn. Massicus raddei
			Aromia bungii
			Monochamus spp.
			Oemona hirta
			Saperda candida
			Xylotrechus spp.
		Chrysomelidae	Diabrotica barberi
ICPR GREAT BRITAIN	Status: APPROVED	19 September 2023	Page 76 of 87

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
			Diabrotica undecimpunctata howardi
Insect			Diabrotica undecimpunctata undecimpunctata
			Diabrotica virgifera zeae
			Epitrix cucumeris
			Epitrix papa
			Epitrix subcrinita
			Epitrix tuberis
			Leptinotarsa decemlineata
		Curculionidae	Anthonomus bisignifer
			Anthonomus eugenii
			Anthonomus quadrigibbus syn. Tachypterellus quadrigibbus
			Anthonomus signatus
			Conotrachelus nenuphar
			Dendroctonus micans
			Euwallacea fornicatus senso lato
			Ips amitinus
			Ips cembrae
			Ips duplicatus
			Ips sexdentatus
			Ips typographus
			Listronotus bonariensis
			Naupactus leucoloma
			Pissodes cibriani
			Pissodes fasciatus
Incont			Pissodes nemorensis
Insect			Pissodes nitidus
			Pissodes punctatus
			Pissodes strobi
			Pissodes terminalis

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
			Pissodes yunnanensis
			Pissodes zitacuarense
			Pityophthorus juglandis
			Platypus apicalis
			Polygraphus proximus
			Premnotrypes spp.
			Pseudopityophthorus minutissimus
			Pseudopityophthorus pruinosus
			Rhagoletis fausta
			Rhynchophorus ferrugineus
			Rhynchophorus palmarum
			Scolytidae spp. (non-European)
			Scolytus morawitzi
			Xyleborus glabratus
		Scarabaeidae	Exomala orientalis syn. Anomala orientalis
			Popillia japonica
	Diptera (flies)	Agromyzidae	Nemorimyza maculosa syn. Amauromyza maculosa
			Liriomyza bryoniae
			Liriomyza huidobrensis
Incost			Liriomyza sativae
Insect			Liriomyza trifolii
		Cecidomyiidae	Aschistonyx eppoi
			Prodiplosis longifila
		Tephritidae	Anastrepha fraterculus
			Anastrepha ludens
			Anastrepha obliqua syn. Anastrepha mombinpraeoptans
			Anastrepha suspensa
			Bactrocera latifrons
			Bactrocera tryoni syn. Dacus tryoni
ICPR GREAT BRITAIN	Status: APPROVED	19 September 2023	Page 78 of 87

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
			Dacus ciliatus
			Dacus dorsalis
			Dacus frontalis
			Dacus punctatifrons
			Dacus tsuneonis syn. Bactrocera tsuneonis
			Dacus zonatus
			Euphranta canadensis syn. Epochra canadensis
			Euphranta japonica syn. Rhacochlaena japonica
			Myiopardalis pardalina
			Neoceratitis cyanescens syn. Pardalaspis cyanescens
			Pardalaspis quinaria syn. Ceratitis quinaria
			Pterandrus rosa
			Rhagoletis cingulata
			Rhagoletis completa
Insect			Rhagoletis indifferens
			Rhagoletis mendax
			Rhagoletis pomonella
			Rhagoletis ribicola
			Rhagoletis suavis
			Strauzia longipennis
			Zeugodacus cucumis
			Zeugodacus cucurbitae syn. Dacus cucurbitae
			Zeugodacus tau
	Hemiptera (aphids, scale,	Aleyrodidae	Aleurocanthus spiniferus
	stink bugs)		Bemisia tabaci
			Trialeurodes abutiloneus
		Aphididae	Toxoptera citricida syn. Toxoptera citricidus
		Cicadellidae	<i>Cicadellidae</i> (known to be a vector of <i>Xylella fastidiosa</i>), such as:

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
			 Draeculacephala Minerva Graphocephala atropunctata
			- Homalodisca vitripennis
			- Xyphon fulgidum syn. Carneocephala fulgida
			Circulifer haematoceps
			Circulifer tenellus
			Hishimonus phycitis
			Scaphoideus luteolus
			Scaphoideus titanus
		Cixiidae	Haplaxius crudus syn. Myndus crudus
Insect		Coccidae	Parasaissetia nigra
1115001			Toumeyella parvicornis
		Diaspididae	Aonidiella citrina
			Lopholeucaspis japonica
			Unaspis citri
		Margarodidae	Margarodes prieskaensis
			Margarodes vitis
			Margarodes vredendalensis
		Phylloxeridae	Daktulosphaira vitifoliae
		Pseudococcidae	Ripersiella hibisci syn. Rhizoecus hibisci
		Triozidae	Bactericera cockerelli
			Trioza erytreae
		Cynipidae	Dryocosmus kuriphilus
		Diprionidae	Gilpinia hercyniae syn. Diprion hercyniae
		Pamphiliidae	Cephalcia lariciphila
	Lepidoptera (butterflies,	Carposinidae	Carposina niponensis
	moths)		Carposina sasakii
		Castniidae	Paysandisia archon
		Crambidae	Diaphorina citri

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
			Neoleucinodes elegantalis
		Gelechiidae	Keiferia lycopersicella
			Tecia solanivora syn. Scrobipalpopsis solanivora
		Lasiocampidae	Dendrolimus sibiricus
		Noctuidae	Helicoverpa armigera syn. Heliothis armigera
			Helicoverpa assulta
Insect			Helicoverpa zea syn. Heliothis zea
			Spodoptera eridania
			Spodoptera frugiperda
			Spodoptera littoralis
			Spodoptera litura
		Notodontidae	Thaumetopoea pinivora
			Thaumetopoea pityocampa
			Thaumetopoea processionea
		Pyralidae	Acrobasis pyrivorella syn. Numonia pyrivorella
		Tortricidae	Acleris gloverana
			Acleris issikii
			Acleris minuta
			Acleris nishidai
			Acleris nivisellana
			Acleris robinsoniana
			Acleris semipurpurana
			Acleris senescens
			Acleris variana
			Choristoneura biennis
			Choristoneura carnana
			Choristoneura conflictana
			Choristoneura fumiferana
			Choristoneura lambertiana
ICPR GREAT BRITAIN	Status: APPROVED	19 September 2023	Page 81 of 87

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
			Choristoneura occidentalis
			Choristoneura orae
			Choristoneura parallela
			Choristoneura pinus pinus
			Choristoneura retiniana
			Choristoneura rosaceana
			Grapholita inopinata syn. Cydia inopinata
			Grapholita packardi syn. Enarmonia packardi
			Grapholita prunivora syn. Enarmonia prunivora
			Platynota stultana
			Thaumatotibia leucotreta syn. Cryptophlebia leucotreta
	Thysanoptera (thrips)	Thripidae	Ceratothripoides brunneus
			Ceratothripoides claratris
			Scirtothrips aurantii
			Scirtothrips citri
			Scirtothrips dorsalis
			Thrips palmi
Molluscs (snails, slugs)	Architaenioglossa	Ampullariidae	Pomacea spp.
Nematode	Dorylaimida	Longidoridae	Longidorus diadecturus
(roundworms)			Xiphinema americanum sensu lato
			Xiphinema americanum sensu stricto
			Xiphinema bricolense
			Xiphinema californicum
			Xiphinema intermedium
			Xiphinema neoamericanum
			Xiphinema rivesi (non-European populations)
			Xiphinema tarjanense
	Rhabditida	Aphelenchoididae	Aphelenchoides besseyi syn. Aphelenchoides oryzae

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
			Bursaphelenchus xylophilus
		Anguinidae	Ditylenchus destructor
			Ditylenchus dipsaci
		Heteroderidae	Globodera pallida
			Globodera rostochiensis
		Meloidogynidae	Meloidogyne chitwoodi
		Pratylenchidae	Hirschmanniella spp. (except Hirschmanniella behningi, Hirschmanniella gracilis, Hirschmanniella halophila, Hirschmanniella loofi and Hirschmanniella zostericola) Nacobbus aberrans
			Radopholus citrophilus
Plant (weeds)	Santalales	Santalaceae	Radopholus similis Arceuthobium spp. (except Arceuthobium azoricum,
Flam (weeus)	Salitalales	Santalaceae	Arceuthobium gambyi and Arceuthobium oxycedri)
Viruses and	Bunyavirales	Fimoviridae	Rose rosette emaravirus syn. Rose rosette virus
viroids		Tospoviridae	Chrysanthemum stem necrosis virus
			Tomato spotted wilt virus
	Mononegavirales	Rhabdoviridae	Citrus leprosis virus
			Strawberry crinkle virus
	Ortervirales	Caulimoviridae	Citrus mosaic virus syn. Citrus yellow mosaic virus
			Strawberry vein banding virus; Strawberry yellow vein-banding virus
	Picornavirales	Secoviridae	Andean potato mottle virus
			Arabis mosaic virus
			Arracacha virus B oca strain
			Blueberry leaf mottle virus
			Cherry leaf roll virus
			Cherry rasp leaf virus
			Peach rosette mosaic virus

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
			Potato black ringspot virus
			Raspberry ringspot virus
			Satsuma dwarf virus
			Strawberry latent ringspot virus
			Tobacco ringspot virus
			Tomato black ring virus
			Tomato chocolate virus
			Tomato marchitez virus
Viruses and			Tomato ringspot virus syn. Tomato ringspot nepovirus
viroids			Tomato torrado virus
	Tymovirales	Alphaflexiviridae	Potato virus X
			Strawberry mild yellow edge virus syn. Strawberry mild yellow
			edge potexvirus
		Betaflexiviridae	Blueberry scorch virus
			Citrus tatter leaf virus
			Cowpea mild mottle virus
			Melon yellowing-associated virus
			Peach mosaic virus
			Potato virus M (Non-European isolates)
			Potato virus S (Non-European isolates)
			Potato virus T
		Tymoviridae	Andean potato latent virus
			Andean potato mild mosaic virus
	Unassigned	Benyviridae	Beet necrotic yellow vein virus
		Bromoviridae	American plum line pattern virus
			Tobacco streak virus black raspberry latent strain syn. Black
			raspberry latent virus
			Potato yellowing virus
			Prunus necrotic ringspot virus

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
		Closteroviridae	Citrus tristeza virus
			Cucurbit yellow stunting disorder virus
			Lettuce infectious yellows virus
			Little cherry virus
Viruses and			Potato yellow vein virus
viroids			Sweet potato chlorotic stunt virus
			Tomato chlorosis virus
			Tomato infectious chlorosis virus
		Geminiviridae	Bean golden mosaic virus
			Beet curly top virus
			Euphorbia mosaic virus
			Florida tomato virus; Tomato mottle virus
			Pepper mild tigre virus
			Squash leaf curl virus
			Tomato leaf curl New Delhi virus
			Tomato yellow leaf curl virus
			Tomato yellow leaf curl Sardinia virus
		Luteoviridae	Grapevine ajinashika virus
			Potato leafroll virus
			Raspberry leaf curl virus
		Ophioviridae	Citrus psorosis virus
		Pospiviroidae	Apple dimple fruit viroid
			Citrus exocortis viroid
			Chrysanthemum stunt viroid
			Coconut cadang-cadang viroid
			Columnea latent viroid
			Pepper chat fruit viroid
		Potyviridae	Cucumber vein yellowing virus
			Potato spindle tuber viroid
ICPR GREAT BRITAIN	Status: APPROVED	19 September 2023	Page 85 of 87

PEST TYPE	ORDER NAME	FAMILY NAME	SPECIES AND GENUS NAME
Viruses and			Plum pox virus
viroids			Potato virus A (Non-European isolates)
			Potato virus V (Non-European isolates)
			Potato virus Y (Non-European isolates)
			Potato virus Yc syn. Potato Y potyvirus stipple streak strain
			Potato virus Y tobacco veinal necrosis strain syn. Potato virus Yn syn. Potato Y potyvirus tobacco
			Potato virus Yo
			Squash vein yellowing virus
			Sweet potato mild mottle virus
			Tomato mild mottle virus
			Tomato planta macho viroid
		Solemoviridae	Blueberry shoestring virus
		Virgaviridae	Tomato brown rugose fruit virus
		Unassigned	Beet curly top virus
			Begomoviruses
			Citrus blight agent
			Strawberry latent C virus syn. Strawberry latent C rhabdovirus
			Non-European viruses and viroids of <i>Cydonia</i> Mill., <i>Fragaria</i> L., <i>Malus</i> Mill., <i>Prunus</i> L., <i>Pyrus</i> L., <i>Ribes</i> L., <i>Rubus</i> L. and <i>Vitis</i> L.

Appendix 3: Sampling schemes of seeds for *Capsicum* spp. and *Solanum lycopersicum*, other than seeds of *Capsicum* spp. varieties which are known to be resistant to the specified pest:

Sampling of seeds for testing shall be carried out under the following sampling schemes depending on the seed lots as referred to in the relevant table of the International Standards for Phytosanitary Measures No 31 Methodologies for sampling of consignments (ISPM31):

- for seed lots less than or equal to 3 000 seeds: application of a hypergeometric sampling scheme able to identify with 95 % reliability a level of presence of infected plants of 10 % or above;
- for seed lots of more than 3 000 but less than 30 000 seeds: application of a sampling scheme able to identify with 95 % reliability a level of presence of infected plants of 1 % or above;
- for seed lots of more than 30 000 seeds: application of a sampling scheme able to identify with 95 % reliability a level of presence of infected plants of 0,1 % or above.

Sub samples shall consist of 1000 seeds maximum for Polymerase Chain Reaction (PCR) methods.

The testing of seeds must be carried out using one of the following methods and the method used must be included in the phytosanitary certificate under the heading "Additional declaration":

- real-time RT-PCR using the primers and probes described in the ISF protocol (2020); or
- real-time RT-PCR using primers and probe of Menzel and Winter (Acta Horticulturae, in press).

Please refer ISPM 31 'Methodologies for sampling of consignments' for more information.